Can Large Pension Funds Beat the Market? Asset Allocation, Market Timing, Security Selection, and the Limits of Liquidity

Aleksandar Andonov, Maastricht University Rob M.M.J. Bauer, Maastricht University, Netspar K. J. Martijn Cremers, Yale University

October 2011

Abstract

We assess and analyze the three components of active management (asset allocation, market timing and security selection) in the performance of pension funds. Security selection explains most of the differences in pension fund returns. Large pension funds in our sample on average provide value to their clients after accounting for all investment-related costs, both before and after risk-adjusting: we find an annual alpha of 17 basis points from changes in asset allocation, 27 basis points from market timing, and 45 basis points from security selection. All three active management components exhibit significant liquidity limitations, which are important in all asset classes, including equity and fixed income. Security selection outperformance is largely driven by momentum trading. Accounting for momentum reduces the security selection alpha by about 72 basis points and offsets most of the positive risk-adjusted returns from market timing and asset allocation changes. Larger funds realize economies of scale in their relatively small allocation to private asset classes, like private equity and real estate. However, in equity and fixed income markets they experience substantial liquidity-related *diseconomies* of scale.

JEL Classifications: G11; G23.

Acknowledgements

We kindly thank CEM Benchmarking Inc. in Toronto for providing us with the CEM database. We thank Frans de Roon for providing factor returns for the Canadian market. For helpful comments and suggestions, we thank Keith Ambachtsheer, David Blake, Jaap Bos, Susan Christoffersen, Alexander Dyck, Piet Eichholtz, Chris Flynn, Mike Heale, Ludovic Phalippou, Peter Schotman, William F. Sharpe, James Xiong and seminar participants at Maastricht University, Rotman ICPM, Dutch Central Bank (DNB) and APG. We gratefully acknowledge a research grant provided by the Rotman International Centre for Pension Management at the Contact authors Rotman School of Management, University of Toronto (ICPM). at a.andonov@maastrichtuniversity.nl, r.bauer@maastrichtuniversity.nl and martijn.cremers@yale.edu.

1. Introduction

Can large, sophisticated investors beat the market? And if so, what investment skills are most prevalent? Can investors outperform by periodically changing asset allocation target weights, by deviating from those in market timing trades, or by selecting securities within asset classes? Do asset allocation and market timing work best using actively managed strategies at higher costs or using a cheaper, passive approach? Are there better opportunities in some asset classes relative to others? What works best: investing internally or using external managers? Finally, are there (dis)economies of scale and liquidity limitations in the answers to these questions? In this paper, we try to address these questions by investigating a unique database of the largest U.S. and Canadian pension funds.

These questions are particularly relevant given the significant aggregate resources devoted to active investing on the one hand (see e.g. French (2008)), and the growing popularity of index funds and index-tracking ETFs on the other hand (see e.g. Cremers and Petajisto (2009) and Cremers, Ferreira, Matos, and Starks (2011)). Such questions have been most intensively investigated in the mutual fund literature. For example, Malkiel (1995), Gruber (1996) and Chan, Chen, and Lakonishok (2002) find that, on average, mutual funds underperform the market by about the amount of expenses charged to investors. More recent studies document evidence that at least some subset of mutual fund managers may have skill. For example, Kacperczyk, Sialm, and Zheng (2008) find that funds that focus on particular industries may outperform, and Cremers and Petajisto (2009) find that funds with high Active Share, i.e., funds whose holdings differ most significantly from those in their benchmark, tend to outperform their index net of all expenses and costs. Kosowski, Timmermann, Wermers and White (2006) find not only that a sizable subgroup of mutual fund managers persist.

Pension funds are large and important investors, playing a vital role in financial markets and influencing general welfare. They are among the largest institutional investors and can influence asset prices and market liquidity through their asset allocation decisions (Allen (2001)). Being responsible for the income of retirees, a poor investment performance of pension funds can not only reduce the wealth and consumption of current and future retirees, but also increase tax burdens if public pension funds fail to meet liabilities (Novy-Marx and Rauh (2011)). The largest defined benefit pension funds are relatively unconstrained, and are able and willing to invest across many different public asset classes (such as equities and fixed income) and private assets (like real estate, private equity and hedge funds), using both active and passive strategies and employing both internal and external investment managers. The long-term liability structure enables pension funds to also invest in the domain of longer-term illiquid assets, in which their vast average size provides significant bargaining power. This makes pension fund performance a particularly rich environment for research, allowing an indepth analysis of all three components of (strategic) portfolio management and of the extent to which all three contribute to performance: asset allocation, market timing and security selection.

Pension funds face an environment that is different from mutual funds. For example, mutual funds are typically much smaller than the pension funds in our sample, and generally have significant

constraints in investing across different (alternative) asset classes. Further, incentives differ substantially. Mutual funds with the best performance receive large cash inflows (see e.g. Sirri and Tufano (1998)). As mutual fund manager pay depends on the size of the assets under management and the relative performance compared to the benchmark, this can create substantial incentives for mutual fund managers to engage in active management or chase short-term performance. However, pension funds' inflows do not depend on performance, but on actuarial and demographic factors, e.g. pension fund maturity or the composition of younger and older workers contributing to and relying on the fund.

We are the first paper, to our knowledge, to provide a comprehensive overview of pension funds' asset allocation, market timing and security selection decisions over two decades, documenting how those decisions relate to their cost structure and their performance.¹ We can do so through access to the unique CEM dataset, comprised of a total of 774 U.S. and Canadian defined benefit pension funds for the period 1990-2008.² This database includes details on each fund's target and actual asset allocation decisions, the self-declared benchmarks for each asset class, and the precise cost structure and performance for all separate asset classes and their benchmarks.

In defined benefit (henceforth, DB) pension funds, plan sponsors have two main investment responsibilities.³ The first involves allocating assets across various asset classes and choosing between active versus passive, and internal versus external management. The second responsibility is to choose and subsequently monitor investment managers. Recent research has focused mainly on the second responsibility, specifically measuring the performance of managers that are hired or fired by plan sponsors (see Goyal and Wahal (2008) and Blake, Timmermann, Tonks and Wermers (2010)). This research design does not allow direct analysis of the total performance of pension funds, since account managers are often hired by more than one pension fund and pension funds typically employ more than one manager. Specifically, this previous literature does not investigate how asset allocation decisions and plan-level choices among managers relate to pension fund characteristics and performance, and typically focuses primarily on equity investments. In this paper, we consider both

¹ A closely related paper is Blake, Lehmann and Timmermann (1999), who investigate the asset allocation and performance of U.K. pension funds throughout the sample period 1986-1994. Their data includes all U.K. funds that maintained the same single, externally appointed fund management group throughout the period. Our data incorporates not only external mandates, but also internal mandates across a more detailed list of asset classes. Another related paper is Brown, Garlappi and Tiu (2010), who consider endowment funds. Endowment funds are similar to pension funds because they also invest simultaneously in equity, fixed income and alternative asset classes. However, the amount of assets under management of pension funds is substantially larger. According to Brown et al. (2010), from 1989-2005, endowment funds had on average \$ 287 million under management, while the mean holdings of U.S. pension funds in our sample is \$ 9,559 million.

 $^{^{2}}$ CEM Benchmarking Inc. (henceforth, CEM) provides services to a larger universe of pension funds, but the U.S. and Canadian samples are by far the largest. Moreover, these funds are based in a similar regulatory environment. Funds in both countries on average have comparable asset allocations: 50%-60% in equities, 30-40% in fixed income and 10% in alternative asset classes.

³ We focus on defined benefit (DB) funds only. In this context, the pension fund's Board makes the asset allocation decisions and is responsible for the eventual performance. In defined contribution (DC) funds, plan sponsors select the menu of available investment options, while each plan member individually is responsible for the actual asset allocation decision. Thus, asset allocation outcomes within DC funds belong more to the literature on individual investors' decision making. Moreover, DC funds usually do not include alternative asset classes in the menu, whereas the alternative asset classes constitute a significant part of the portfolio of a typical DB fund.

responsibilities of plan sponsors, asset allocation policy and its relation to overall pension fund performance measured at the *total fund level* (i.e., not at the mandate or manager level). The overall pension fund performance incorporates the performance in equity, fixed income and alternative asset classes such as real estate, private equity and hedge funds. Pension funds in our sample have both internal and external managers, and combine both active and passive strategies.

Our main findings are seven-fold, collectively suggesting strong evidence for the ability of the pension funds in our sample to outperform, though subject to significant liquidity limitations. First, we document the asset allocation decisions and cost structures of large U.S. and Canadian pension funds. Pension funds make similar asset allocation decisions, with a typical pension fund in our sample investing around 55% of the assets in equity, 35% in fixed income and 10% in different alternative asset classes, and only limited cross-sectional dispersion. Equity and fixed income holdings mainly consist of domestic assets, with international diversification increasing over time.⁴ Real estate is the most important alternative asset class in both countries, accounting for 4% of U.S. and 3% of Canadian funds' total assets under management, with 77% (64%) of the pension funds in U.S. (Canada) investing in real estate over the 1990-2008 period. More than 80% of the assets of the pension funds in our database are invested in active mandates and this pattern persists in all asset classes and across time. This helps explain the large cross-sectional differences in returns across pension funds. Only 15% of assets are managed internally, mostly by the largest funds.

Although U.S. funds are on average larger than Canadian funds, this does not result in lower costs. The total investment costs of U.S. pension funds are on average 35.25 basis points per year, whereas Canadian funds exhibit costs of 25.65 basis points. This is somewhat surprising, because in general we find strong evidence of economies of scale in costs, with larger funds having lower costs per dollar invested.⁵ The costs difference could imply that Canadian funds are better governed or that especially the larger U.S. funds have a potential to further reduce their investment costs by more strongly exercising negotiation power.

Investment costs are stable during the first half of our sample, but continuously increase after 1999. This trend is largely due to the higher allocation to alternative assets, especially hedge funds, which have much higher costs. In 2008, the average cost of U.S. funds was 49.72 basis points per year, while Canadian funds paid 33.92 basis points for their investments. Over the entire period, the most expensive asset class is private equity (average cost of 280 basis points per year⁶), while the least expensive classes are fixed income and cash (14-19 basis points).

⁴ U.S. (Canadian) funds' investments in domestic equity represented 89% (74%) of total equity holdings in 1990, while in 2008 their share reduced to 64% (43%), with most of the shift going into global equity funds.

⁵ Even though larger funds have more negotiation power and can capitalize on economies of scale, our findings indicate that Canadian funds have lower costs on the fund level, but also separately in all major asset classes. The difference in costs is also not due to more passive or internal management among Canadian funds, because pension funds in both countries manage on average around 80% of their assets actively and externally (see also Bikker, Steenbeek and Torracchi (2010)).

⁶ This estimation understates the actual costs of investing in private equity (see Phalipou (2009) for a detailed analysis of private equity fees). It captures the management fees, but the performance fees are subtracted directly from the returns. Nevertheless, in the calculation of private equity net returns both management and performance fees are deducted.

Second, we decompose pension fund returns in three components (asset allocation, market timing and security selection). The first component, asset allocation, is calculated in two ways. When comparing the importance of asset allocation, market timing and security selection for explaining net performance variability, we define the asset allocation return component as a deviation in the strategic (target) asset allocation policy from the average asset allocation policy of all funds in one year. We do so in order to conform as closely as possible to Xiong, Ibbotson, Idzorek and Chen (2010). When we evaluate pension fund performance, asset allocation return is calculated using the changes over time in each fund's ex-ante declared 'target' asset allocation weights times the self-declared benchmark returns of the different asset classes. For each separate asset class within each fund, we observe the self-declared benchmark as well as the return on these benchmarks. Asset allocation performance evaluation thus compares the performance of the change in target weights over last year, relative to not changing last year's target weights. The second component is market timing (or tactical asset allocation), defined as the difference between target and actual (realized) weights. Market timing thus captures the performance related to overweighting or underweighting particular asset classes, relative to the target weights in that year.⁷ The third and last component is security selection, corresponding to benchmarkadjusted net returns or the difference between realized net returns and benchmark returns for a given asset class. This component captures the returns due to picking securities and timing industries and styles within an asset class. All three components are measured after accounting for all investment costs.

Net performance variability comes mainly from security selection: 45-55% in the U.S. and 48-58% in Canada. Asset allocation decisions explain only 35-41% of the return differences in the U.S. and even less in Canada (25-34%), with the balance attributed to market timing.⁸

Third, we document that pension funds are, on average, able to beat the market or their self-declared benchmarks, both before and after risk-adjusting for equity market, size, value, liquidity and fixed income market factors. Interestingly, they can do so in all three components of active management. Pension funds show skill with respect to setting asset allocation target weights (17 basis point annual alpha), the timing of asset allocation decisions (27 b.p. annual alpha), and derive an even larger positive alpha resulting from security selection decisions (45 b.p. per year).

Fourth, we offer particular interpretations of the security selection results. For U.S. funds, the positive alpha from security selection, 28 basis points per year (marginally significant only with a z-statistic of

⁷ For instance, if a fund's strategic weight for U.S. equity is 60%, but the realized weight is 65% (and say for U.S. fixed income the strategic weight is 40% and the realized weight is 35%), the market timing components for U.S. equity (fixed income) equals +5% (-5%), multiplied by the relevant benchmark return. The main difference between asset allocation and market timing is horizon. Asset allocation target weights change less frequently: many fund-years observations show no change in asset allocation. Market timing is shorter-term oriented, with very few funds having no difference between the target and the actual weights in any given year.

⁸ Xiong, Ibbotson, Izdorek and Chen (2010) decompose the returns of mutual funds in a similar fashion. Their results show that differences in asset allocation policy and active portfolio management are equally important for mutual funds. Relative to mutual funds, pension funds have a considerably greater opportunity to invest in multiple asset classes and to change investment allocations strategically. Most mutual funds are limited to invest in either equity or fixed income, and 'balanced' mutual funds typically only include equity and fixed income investments but no alternative asset classes. Therefore, *ex ante* the asset allocation policy would seem to be more important for pension funds than mutual funds, such that our results are surprising.

1.70) is fully driven by momentum. The momentum factor captures the difference in returns between a portfolio of stocks with high prior one-year returns (winners) and a portfolio of stock with low prior returns (losers). Adding the momentum factor to the risk-adjusting model, U.S. funds security selection performance turns negative at -107 basis points a year, indicating that momentum strategies deliver about 135 basis points a year.

Canadian funds exhibit a security selection alpha of 83 basis points per year (z-statistic of 2.98), all of which we cannot ascribe to active management nor to momentum but rather to the "Nortel" effect.⁹ Adjusting for the "Nortel" effect, the security selection component of Canadian funds equals -4 basis points per year (or -21 basis points a year controlling for the momentum factor, though neither is significant).¹⁰

Blake, Lehmann and Timmermann (1999) find that the cross-sectional return variation among U.K. pension funds in the period 1986-1994 is also dominated by the security selection component. However, contrary to our findings their results indicate negative returns from market timing, attributed to negative timing returns within foreign equity (see also Timmermann and Blake (2005)). The security selection returns of U.K. funds are positive, but not always significant. One important difference in the construction of the market timing return component is that we have access to the strategic asset allocation weights and self-determined benchmarks, whereas Blake, Lehmann and Timmermann (1999) use one benchmark index per asset class as a return proxy for all pension funds and estimate the strategic weights based on the trend in realized weights. Another difference is that we look not only at the external mandates, but also at the internal mandates across all asset classes. Moreover, we do not require that a single external manager is employed during the entire sample period.

Fifth, in the last step of our analysis, we relate the risk-adjusted returns (on a total fund or asset class level) for the changes in asset allocation, market timing and security selection components to the total size and liquidity of the funds' holdings, the size and liquidity of the investments in a particular asset class, the investment costs and the investment style. The investment style reflects whether assets are managed internally or externally, and whether the assets are managed passively or actively.

The relationship between asset size and performance is not uniform and depends on the asset class and investment style. Larger funds realize economies of scale in alternative asset classes, especially real

⁹ The "Nortel Effect" refers to the fact that in July 2000 Nortel Company constituted 36% of the S&P/TSX Composite index. Nortel's return was 69% from January to July 2000, but the overall return for the year 2000 was -33.8%. The volatile returns on Nortel created significant differences between returns on the TSE300 Composite Index (7.4% in 2000) and the capped version of the same index (19.1% in 2000). In other years, there are only minor differences between the two versions of TSE index. The investment decisions of Canadian funds concerning Nortel resulted in a substantial outperformance of the domestic equity benchmark in 2000. Following the index in 2000 was dangerous, because a portfolio with 36% invested in one company cannot qualify for 'diversified investing', as it is exposed to substantial idiosyncratic risk.

¹⁰ The question of whether momentum is a priced risk factor (or can be explained by risk) is clearly debatable. However, most literature suggests that it cannot be explained by exposure to systematic risk factors (see e.g. Jegadeesh and Titman (1993, 2001), Lee and Swaminathan (2000), Cooper, Gutierrez and Hameed (2004) and Cremers and Pareek (2011)). Even the papers arguing for a risk-based interpretation acknowledge that momentum cannot be mostly or completely explained by risk (see e.g. Grundy and Martin (2001) and Lu and Zhang (2008)).

estate, but experience *dis*economies of scale in public equity and fixed income markets. These diseconomies of scale are mainly driven by liquidity constraints. Internal management is associated with improved security selection performance.

Higher costs are generally related to lower performance, but only after controlling for momentum. Further, the impact of investment costs on performance varies between asset classes. For instance, the negative relationship on the total fund level is mainly driven by the negative relationship between costs and performance in equity and alternative assets. Particularly in private equity and real estate portfolios, investment costs have a strong negative effect on net returns. We find some evidence that asset allocation performance is best achieved using passive rather than active management, which is related to liquidity as well, as passive investing generally means more liquidity.

Sixth, all three components of active management exhibit liquidity limitations, which seem quite important even for asset classes such as equity and fixed income. Shifts in the strategic asset allocation towards more illiquid assets hurt the performance of larger funds relative to smaller funds. Similarly, smaller funds can more effectively do market timing without distorting market prices. Finally, the security selection performance of larger funds seems particularly constrained by liquidity, with significant economic effects: increasing liquidity by lowering the liquidity beta by 10 percentage points is associated with an improvement of the alpha of funds at the 75th size percentile by 15 basis points per year more than the improvement of the alpha of a fund at the median size percentile.

Seventh and finally, we document strong performance persistence for both market timing and security selection using annual quintile rankings. Funds are more likely to end up in a better performing quintile next year, if they also do so this year, and they are more likely to perform worse in the ranking next year if they performed relatively poorly this year.

The empirical conclusions are not likely to be influenced by a self-reporting bias. Results from a Cox proportional hazard model show that the database does not seem to suffer from this bias with respect to costs and returns, though larger funds are more likely to survive in the CEM database. The database is most inclusive for Canada: CEM covers approximately 30-40% of the total assets managed by U.S. DB pension funds and 80-90% by Canadian funds. Further, sample selection and survivor issues appear *ex-ante* to be greater for the U.S. sample (due to lower coverage). The general consistency of results across both countries strengthens our conclusion that the database does not suffer from self-reporting biases related to performance.

Our finding that smaller fund or mandate size results in better security selection returns in equity were already documented for this same sample in Bauer, Cremers and Frehen (2010), who exclusively study the performance of the domestic equity portfolios of U.S. pension funds only. It is also similar to findings of Lakonishok, Shleifer and Vishny (1992), who showed that equally-weighted equity returns of funds are higher than value-weighted returns in the period 1983-1989. In addition to costs advantages, increased scale can be expected to have a positive impact on the level of expertise in the selection and monitoring of investment managers. However, *dis*economies of scale related to organization and liquidity problems have been found among mutual funds (Chen, Hong, Huang and

Kubik (2004)), among private equity companies (Lopez-de-Silanes, Phalippou and Gottschalg (2010)) and among hedge funds (Fung, Hsieh, Naik and Ramadorai (2008)). Our results point mostly towards larger funds being constrained by liquidity. In doing so, we borrow the methodology and again confirm the results in Bauer, Cremers and Frehen (2010) for domestic equities of U.S. funds, but in our paper give those also at the total fund level, for Canadian funds as well as for asset classes other than U.S. equities.

Our results partially contradict the existence of economies of scale in pension fund management as discussed in Dyck and Pomorski (2011). The difference in results can largely be explained by a difference in methodology: we analyze not only the non-risk-adjusted returns, but we also risk-adjust fund performance for factor returns, investigate the importance of momentum and control for fund fixed effects.¹¹ Dyck and Pomorski (2011) do not risk-adjust returns and focus on specifications without fund fixed effects and without controlling for momentum.¹² In our view, risk-adjustment is critical for performance evaluation and merely benchmark-adjusting is insufficient, as is borne out by our results.¹³ At the fund level, for example, we find that evidence that larger U.S. funds do better than smaller U.S. funds (see e.g. Dyck and Pomorski (2011)) disappears once we risk-adjust returns. Evidence that larger Canadian funds do better than smaller Canadian funds can be completely explained by larger Canadian pension funds being more active in pursuing momentum strategies than smaller Canadian pension funds. Specifically, after risk-adjusting, we only find a positive association between alpha and size if we do not control for momentum, and then only for Canadian but not for U.S. pension funds. We generally do not find economies of scale in equity and fixed income, but we confirm Dyck and Pomorski's finding that larger funds perform better in private equity and especially real estate.

Our results show that large funds that manage most of their assets internally improve their performance compared to peers with similar size but mostly external managers. Dyck and Pomorski (2011) also conclude that internal management improves pension fund performance mainly through cost savings. Internal management can reduce potential agency conflicts from multiple layers (Lakonisok et al. (1992)) and also results in lower investment costs. However, internal management is a realistic option only for larger funds that can devote sufficient resources to establishing an internal asset management department.

The empirical results finally suggest that larger funds can assert more negotiation power in alternative asset classes, which may lead to greater access to the best investment opportunities at lower costs. Larger funds can devote more resources to monitor private equity and real estate investments. The

¹¹ Robustness of our risk-adjusted results can be checked by comparing Appendix Table A.2 with Table 7.

¹² In Appendix Tables A.7 and A.10 we replicate part of Dyck and Pomorski (2011) findings of economies of scale among pension funds before risk-adjusting.

¹³ Additionally, Dyck and Pomorski (2011) transform all returns and holdings of Canadian pension funds in U.S. dollars using the end-of-year exchange rate. We believe that this transformation introduces unnecessary time series variation in Canadian funds' returns and assets size. Domestic assets constitute the major part of Canadian funds' portfolio and most of the pension funds hedge the exchange rate risk when investing in international markets. Hence, the returns and holdings of Canadian funds do not fluctuate together with the exchange rate between the U.S. and Canadian dollar.

largest funds even establish internal or "at-arms-length" operating private equity and real estate divisions. The importance of lower cost is especially pronounced among U.S. funds investing in private equity and Canadian pension funds investing in real estate.

The paper proceeds as follows. Section 2 describes the CEM dataset and considers possible selfreporting biases. Section 3 explains the methodology to decompose fund returns into asset allocation, market timing and security selection components, and to measure the importance of each component in explaining the differences in returns between pension funds. Section 4 describes the methodology employed to measure the fund (and asset class) risk-adjusted performance and its relation to fund characteristics, and presents the empirical results. Section 5 briefly discusses the persistence in pension fund performance. Concluding comments are provided in section 6.

2. Characteristics of the CEM database

CEM Benchmarking Incorporated (henceforth CEM) collects Canadian and U.S. defined benefit pension fund data through yearly questionnaires.¹⁴ The CEM database contains detailed information on pension fund holdings, costs, benchmarks and returns on the fund level and per asset class. Table 1 illustrates the time trend in the number of funds reporting to CEM. In the period 1990–2008, a total of 774 U.S. and Canadian pension funds have reported to CEM. The number of funds reporting is lower in the first three years of the database formation, but afterwards it is stable over time. The main motive for pension funds to enter the database is to benchmark their costs against peers based on total fund size and total holdings in particular asset classes. Funds sometimes decide to stop submitting the questionnaires to CEM for various reasons, such as termination of the service due to costs savings, mergers, acquisitions and bankruptcies of the underlying corporations etc. As reporting to CEM is voluntary, the data is potentially vulnerable to self-reporting bias. Bauer, Cremers and Frehen (2010) address the self-reporting bias by matching the CEM data with the Compustat SFAS data and testing whether the decision to stop reporting is related to the overall fund performance. Their results indicate that there is no evidence of a self-reporting bias related to performance in the exiting and entering years.

Here, we address the self-reporting problem by constructing a Cox proportional hazard model. We test whether the decision of a particular pension fund to exit the database is related to its returns, its costs or its size. The event of interest is the decision of the pension funds not to report to CEM in a given year. In the Cox hazard model, we treat each fund re-entry as a new fund, which explains why the number of units in Table 2 (column 1) is higher than the total number of funds presented in Table 1 (column 2). The results in Table 2 indicate that fund size ("log(Size)") has the strongest effect on the fund's exit rate. Size in this case refers to the total holdings (asset under management) by the pension fund. For example, a hazard ratio of 0.7483 (-6.81) means that an increase by one unit in log(size)

¹⁴ Other papers studying pension fund performance using the CEM database are French (2008), Bauer, Cremers and Frehen (2010) and Dyck and Pomorski (2011). The CEM database also includes information of pension funds in Europe and Australia, and includes both defined benefit and defined contribution plans.

leads to a decrease of 25.17% (100% - 74.83% = 25.17%) in the exit rate (see first row in panel A). Panels B and C show that the results from the Cox proportional hazard model for all funds are consistent with the findings in the U.S. and Canadian subsamples.

Overall, Table 2 shows that smaller funds are more likely to exit the CEM database. This is consistent with the idea that specialized benchmarking services provided by CEM are more relevant and costeffective for larger funds. Further, we relate the fund exit rate to pension fund net returns, benchmark returns and benchmark-adjusted returns. Net returns are obtained after subtracting total costs from gross returns. Benchmark returns are calculated using the benchmarks reported by pension funds for every asset class in which they invest. Every year, CEM asks funds to report the exact definition of the benchmark they employ, as well as the return on that benchmark for every asset class in which a fund has holdings. We specify benchmark-adjusted net returns as gross fund returns minus costs, and minus benchmark returns indicate that funds are more likely to stop reporting in years that financial markets perform well. For instance, the hazard ratio of 1.0220 (t-statistic of 1.86) on net returns in panel B indicates that a one-percentage point increase in net returns increases the exit rate by 2.20%. Hazard ratios of benchmark-adjusted net returns are always insignificant, so we can conclude that exit events are not related to funds underperforming or outperforming their benchmark.¹⁵ Hence, we find no evidence that the CEM database suffers from self-reporting bias related to performance.

Total costs are somewhat negatively related to the exit rate of U.S. funds. The hazard ratio of 0.9915 (t-statistic of -1.76) in Panel B indicates that an increase in costs by one basis point results in 0.85% decrease in the exit rate. Funds with higher costs may benefit more from the cooperation with CEM, because the company is specialized in advising on costs. Overall, the self-reporting tests suggest that CEM suits better the interests of larger funds, but the dropping rate is not related to benchmark-adjusted performance and only marginally to the cost level.

Funds included in the CEM database cover a substantial share of the pension fund assets under management and market capitalization in the U.S. and Canada. For example, Canadian pension funds' holdings in Canadian equity represent approximately 11.8% of the total market capitalization of Toronto Stock Exchange (TSX) in 2008. Over the 1990-2008 period, Canadian funds included in the CEM database account for approximately 80-90% of the asset under management by Canadian pension funds. U.S. funds in the same time period comprised around 30-40% of the asset under management by U.S. pension funds. In 2008, the holdings in U.S. equity of U.S. pension funds included in the CEM universe represent 6.5% of the market capitalization of the NYSE, NASDAQ and AMEX and their fixed income holdings are equal to about 2% of the total outstanding U.S. bond market debt in 2008.

The U.S. funds in our sample are significantly larger than the Canadian funds (see Table 3). The average and 75% percentile of fund size equal 9.6 billion USD and 8.1 billion USD for U.S. funds,

¹⁵ In Appendix Table A.11 we sort the funds into five quintiles based on their market timing and security selection returns. The percentage of U.S. and Canadian funds exiting the database is similar across all quintiles, i.e. top performers have very similar exit rates as the worst performers.

respectively, versus 4.4 billion CAD and 2.6 billion CAD for Canadian funds. The positive skewness indicates that the CEM universe consists of several very large and many smaller funds.

We can distinguish the following asset classes, with their average weights for U.S. / Canadian funds, respectively: equity (58% / 54%), fixed income (31% / 38%), cash (2% / 3%), real estate (4% / 3%), private equity (2% / 1%) and other (2% / 1%). Panel A of Figure 1 presents the time trend in the allocation to equity, fixed income, cash, real estate, private equity and other assets among U.S. funds. Panel B presents the same information for Canadian funds. In the period 1990-2000, allocations to equity increase in both countries. The most important alternative asset class for both U.S. and Canadian pension funds is real estate.¹⁶ U.S. funds allocate a higher percentage of their assets to private equity compared to Canadian funds.¹⁷ "Other" presents a heterogeneous category consisting of assets, which separately constitute only a minor part of pension funds holdings. It encompasses the following asset types: tactical asset allocation (TAA) mandates, infrastructure, hedge funds, commodities and natural resources, which has been growing in importance. In 2000, fewer than 3% of funds have hedge fund investments, while 43% of U.S. funds and 27% of Canadian funds do in 2008 (hedge fund investments in 2008 represent 3.33% of total assets for the U.S. and 1.65% of total assets in Canada).

To summarize, pension funds seem to display large degrees of herding in the asset allocation decisions. A typical pension fund in our sample invests around 55% of the assets in equity, 35% in fixed income and 10% in different alternative asset classes, with only limited cross-sectional dispersion. This is consistent with the observation of Lucas and Zeldes (2009). Using a sample of U.S. public pension funds, they show that variation in the equity share in the funds' portfolios is not explained by the percentage of active participants, differences in funding ratios and other variables suggested by theory to be relevant for asset allocation policy.

Bauer, Cremers and Frehen (2010) document a negative relationship between fund size and costs for investing in U.S. equities. This negative relationship is robust to the investment style, i.e. it is not driven by the higher proportion of passive and internal investments among larger funds. Larger funds are able to negotiate lower fees for external mandates and organize more cost-efficient internal mandates. Considering the total fund level here, the negative relationship between fund size and costs exists within both Canada and the U.S. However, Canadian funds have significantly lower costs than U.S. funds, even though they are much smaller. Panel B of Table 3 demonstrates that Canadian funds have lower costs across all asset classes, except "other". The higher costs of Canadian funds in the "other" category are largely due to the higher allocation to infrastructure (which is relatively expensive). Apparently, Canadian pension funds are able to negotiate lower fees for investing in equity, fixed income, cash, real estate and private equity. On a total fund level, pension funds from Canada paid on average 25.65 basis points, whereas U.S. funds spent around 35.25 basis points for

¹⁶ The real estate category includes assets invested in direct real estate holdings, segregated real estate holdings, real estate limited partnerships and real estate investment trusts (REITs).

¹⁷ Private equity includes venture capital, LBO and energy partnerships, as well as equity or fixed income investments in turnarounds, start-ups, mezzanine, and distress financing.

investing in the same asset classes (in the 1990-2008 period). This cost difference is not driven by a higher allocation to passive mandates, which are by construction cheaper than active mandates. Surprisingly, U.S. funds even have a slightly higher allocation to passive and thus less expensive mandates in equity than Canadian funds (27.18% versus 15.67%). Furthermore, the difference is present in every year and is not due to outliers, as indicated by lower 25th percentile and 75th percentile cost values for Canadian funds in Table 3.

Figures 2 and 3 plot the time variation in asset allocation within equity, fixed income and alternative asset classes of U.S. and Canadian Funds. Panel A in both figures shows that pension funds invest the majority of their equity holdings in domestic stock markets, with international diversification increasing over time. For instance, U.S. funds invested 89.47% of their total equity holdings in U.S. markets in 1990, while this percentage decreased to 64.23% in 2008. The same trend can be noticed among Canadian funds in Figure 3 (Panel A). The allocation to Canadian equity decreases from 73.84% in 1990 to 42.53% in 2008.

In both countries, the decrease in domestic equity is reallocated to either an EAFE mandate, capturing about 18% and 22% of the equity holdings of U.S. and Canadian funds respectively, or a global equity mandate.¹⁸ For example, the allocation to these global mandates is 14.67% in 2008 for U.S. funds.

Panel B in Figures 2 and 3 plots the time variation of allocation to various fixed income asset classes. The focus on domestic investments is even more striking here. In 1990, U.S. funds held 96.64% of their fixed income investments in the U.S. market (and 99.36% for Canada), with only very limited international diversification since then. For instance, the allocations to EAFE, Emerging Markets and Global fixed income mandates remain low and stable over the 1990–2008 period for both U.S. and Canadian funds (less than 8% combined).

Overall, the asset allocation policy of pension funds in equity is similar to the policy of endowment funds described in Brown, Garlappi and Tiu (2010). However, pension funds allocate a higher proportion of their assets to fixed income securities than endowments (20-25%). Furthermore, the most important alternative asset class category for endowments is hedge funds, whereas pension funds focus more on real estate and private equity investments.

3. Decomposing pension fund returns

In addition to realized (actual) asset allocation weights, CEM also provides information on the pension fund target policy weights, which are determined by the pension funds' Boards. The changes in policy weights from year t-1 to year t show how pension fund strategic allocations evolve over time. Table 4 shows that U.S. funds modified their strategic allocation by adding more private equity and other alternative assets at the expense of fixed income and cash. Canadian funds also reduced the target weights for fixed income and cash, but increased mainly the equity target weights. The realized (or actual) weights vary around the target asset allocation weights. Table 4 further presents information on

¹⁸ EAFE mandates refer to equity investments in Europe, Australasia and Far East (developed countries). Most global mandates use the ACWxUS ("All countries in the World excluding US") benchmark.

the differences between the reported target weights and realized (actual) weights, which are close to zero on average, but with (averaged across time) cross-sectional standard deviations of 1.5% - 5.5%.

To estimate and compare the importance of the asset allocation, market timing and security selection, we follow the methodology of Xiong, Ibbotson, Idzorek and Chen (2010) and Brown, Garlappi and Tiu (2010). A fund's total return can be decomposed into four components: (1) the average policy return, (2) the asset allocation policy return in excess of the average policy return, (3) market timing and (4) security selection or active management returns. The total fund return is the weighted return on all assets, in which one fund has holdings, net of all expenses and fees.

Our measure of the average policy return, PR_t , is the average of the equally-weighted policy return for a given year of all the funds in the database. The policy return for every fund is calculated using the target policy weights in a given year times the benchmark returns.

$$PR_{t} = \frac{\sum_{i=1}^{N} \sum_{j=1}^{M} w_{i,j,t}^{AA} r_{i,j,t}^{BM}}{N_{t}}$$

where $w_{i,j,t}^{AA}$ is the policy weight of fund *i* for asset class *j* in year *t*, $r_{i,j,t}^{BM}$ is the benchmark return on asset class *j* for fund *i* in period *t*, and N_t represents the number of funds in year *t*.

Xiong, Ibbotson, Idzorek and Chen (2010) argue that overall market movements dominate the timeseries analysis of total returns in time series regressions, accounting for about 80-90% of the total return variation and obscuring the contributions of asset allocation, market timing and security selection. Cross-sectional regressions naturally remove the influence of market movements, essentially resulting in the same analysis as using excess market returns (see Xiong et al. (2010)). In our methodology, market movements are captured by the average policy return (PR_t).

Following Brown, Garlappi and Tiu (2010), we decompose the total returns of pension funds in excess of the average policy return (or in excess of the market) into three components: (1) asset allocation, (2) market timing and (3) security selection. Formally, let $R_{i,t}$ be the realized return on fund *i* at the end of year *t*, $w_{i,j,t}$ the actual realized weight of fund *i* in asset class j = 1, ..., M and year *t*, and $r_{i,j,t}$ the realized net return on the asset class *j* for the year *t* by fund *i*, then

$$R_{i,t} - PR_t = \sum_{j=1}^{M} (w_{i,j,t}^{AA} r_{i,j,t}^{BM} - PR_t) + \sum_{j=1}^{M} (w_{i,j,t} - w_{i,j,t}^{AA}) r_{i,j,t}^{BM} + \sum_{j=1}^{M} w_{i,j,t} (r_{i,j,t} - r_{i,j,t}^{BM})$$
$$\equiv R_{i,t}^{AA} + R_{i,t}^{MT} + R_{i,t}^{SS}.$$

The first component, $R_{i,t}^{AA}$, indicates the return from the asset allocation policy in excess of the average policy return. $R_{i,t}^{MT}$ captures market timing, estimated as returns due to the difference between actual realized weights and target asset allocation weights in different asset classes. The last component, $R_{i,t}^{SS}$, measures returns from security selection, estimated as the difference between the realized net returns and the benchmark returns. Hence, the security selection component is equivalent to benchmark-adjusted net returns and accounts for all returns that are not attributable to policy decisions or timing decisions. The market timing term will account for returns due to changes in the weights *between* asset

classes, not *within* a mandate. For instance, it will capture returns due to a higher allocation to equity at the expense of bonds, or returns due to a higher allocation to domestic equity instead of an EAFE mandate. However, the market timing component will not capture returns due to overweighting particular industry sectors within the U.S. equity mandate.

Related to our measure of market timing, pension fund Boards in practice usually determine not only the target asset allocation mix, but also the range in which these weights can fluctuate (bandwidths). For example, a pension fund board can decide that their policy weight to equity is 60% of assets under management, with a range of 57% to 63%. If the allocation to equity goes below 57% or above 63%, the fund will rebalance its portfolio in order to restore the strategic asset allocation mix. Thus, the differences between realized and target weights can result from market movements and active rebalancing decisions of investment managers. The market timing component will capture both causes. Table 5 summarizes the decomposition of the variation in total net return in excess of the average policy return ($R_{i,t} - PR_t$), in terms of the average R-squared. It shows the average contribution of each component to the variation in the difference between the total net return and the average policy return. This table reports time-series and cross-sectional R-squared summary statistics on the fund level, incorporating all assets. Panel A displays summary statistics from the cross-sectional distribution of R-squared statistics obtained from performing the following regression for each pension

$$R_{i,t} - PR_t = \alpha_i + \beta_i R_{i,t}^{\kappa} + \varepsilon_{i,t} , \qquad i = 1, \dots, N$$

where $R_{i,t}$ is the net return of pension fund *i* at time *t*, and PR_t is the average of equally weighted policy return for year *t*. $R_{i,t}^k$ can refer to the asset allocation return component $R_{i,t}^{AA}$, the market timing component $R_{i,t}^{MT}$ or the security selection component $R_{i,t}^{SS}$.

The second part of Table 5 reports summary statistics from the time-series distribution of R-squared from the 19 (1990–2008) cross-sectional regressions:

$$R_{i,t} - PR_t = \alpha_i + \beta_i R_{i,t}^k + \varepsilon_{i,t} , \qquad t = 1, \dots, T$$

where k = AA, MT, SS. At least five data points per fund are required to run each time-series or crosssectional regression. Hence, in the regressions on all funds we include 348 funds, of which 217 are U.S. funds and 131 Canadian. Our results are robust to using cutoff thresholds with more or less than five data points (see appendix Table A.1 for results using at least 4, 7 and 9 data points).

We first consider the results for the "All Funds" universe. The asset allocation policy has the highest explanatory power, accounting for 51% (cross-sectional) to 60% (time series) of the variation in returns between funds. We expected these results, since U.S. and Canadian funds have very different asset allocation policies, as presented in Figures 2 and 3. U.S. and Canadian funds hold the majority of their assets in domestic equity and domestic fixed income mandates.¹⁹ This explains the high explanatory power of the asset allocation decision. Table 5 further shows that the security selection

¹⁹ For instance, Canadian funds invest on average 57% of their total equity holdings in the Canadian market. U.S. funds do not have a separate "Canadian equity" asset class.

component accounts for 32-34% of the variations in excess policy return. Finally, the market timing component explains 5-18% of the return differences between all funds.

We explicitly analyze the single-country subsamples because the results for the all funds universe are mainly driven by the strong home bias in both countries. When U.S. and Canadian pension funds are analyzed separately, results are different. Security selection now has the highest explanatory power, accounting for 45-55% of the variation in excess policy returns in the U.S. and 48-58% in Canada. The explanatory power of the market timing component remains around 4-18% in both countries. The asset allocation decision explains only 35-41% of the return differences of U.S. funds and even less in Canada: 25-34%. These findings are in line with the research on mutual funds by Ibbotson and Kaplan (2000), who find that the asset allocation policy has a cross-sectional R-squared of around 40%. Brown, Garlappi and Tiu (2010) have even more pronounced results and find that the security selection component is responsible for most (74.69%) of the cross-sectional variation in endowment returns. The research on U.K. pension funds by Blake, Lehmann and Timmermann (1999) finds fairly low probabilities of individual fund asset allocations remaining above or below the industry average weight each year, also implying possible herding behavior.

Asset allocation has higher explanatory power within the U.S. as compared to Canada, partly because U.S. funds have more specific benchmarks and lower aggregation levels within equity investments. First, U.S. funds can choose different benchmarks for domestic equity (e.g. Russell 3000, S&P 500, Russell 2000 and Wilshire 5000), whereas the vast majority of Canadian funds use the S&P/TSE300 as their benchmark. Second, the CEM reporting structure allows U.S. funds to report large and small cap domestic equity investments separately (which is not available for Canadian funds).

As pension funds are relatively unconstrained and can invest in all asset classes, it may be surprising that security selection has a higher explanatory power than asset allocation. These results are driven mainly by two causes. First, while U.S. and Canadian funds have an opportunity to choose from multiple asset classes, most of them end up with a conventional asset allocation mix. The typical pension fund in our sample invest between 55-60% of the asset in equity, 30-35% in fixed income and around 10% in alternative assets, with a substantial proportion of the equity and fixed income assets held in domestic markets. Second, U.S. and Canadian funds employ a large degree of active management. During the 1990–2008 period, Canadian and U.S. funds held on average 84% and 73% of their equity holdings in actively managed mandates, respectively (with a similar picture for fixed income, (81% and 85%, respectively)). Active management can lead to substantial differences in returns between pension funds. Ibbotson and Kaplan (2000) present consistent evidence that active management decreases the cross-sectional R-squared of the asset allocation components of mutual funds.

4. Pension Fund Characteristics and Performance

In this section, we discuss whether asset allocation, market timing and security selection decisions result in outperformance or underperformance of pension funds. To evaluate pension fund

performance, we employ two different methodologies: the Swamy (1970) random coefficient model as well as Fama-MacBeth (1973) regressions.

4.1 Methodology

To estimate and evaluate the asset allocation skills of pension funds, we look at the yearly changes in pension fund strategic asset allocations.²⁰ The returns due to such changes are estimated as the difference between pension fund *i* policy (i.e., target) weights for asset class *j* in year *t* compared to year *t*-1 multiplied with the benchmark return of that asset class at time *t*:

$$R_{i,t}^{CAA} = \sum_{j=1}^{M} (w_{i,j,t}^{AA} - w_{i,j,t-1}^{AA}) r_{i,j,t}^{BM}$$

We define market timing returns of fund *i* in year *t* ($R_{i,t}^{MT}$) as the pension fund return due to a deviation from strategic asset allocation policy:

$$R_{i,t}^{MT} = \sum_{j=1}^{M} (w_{i,j,t} - w_{i,j,t}^{AA}) r_{i,j,t}^{BM}$$

Following the previous section, our security selection return component $(R_{i,t}^{SS})$ of fund *i* in year *t* represents benchmark-adjusted net returns, i.e. returns that are due to deviation from self-declared benchmarks within the particular asset class:

$$R_{i,t}^{SS} = \sum_{j=1}^{M} w_{i,j,t} (r_{i,j,t} - r_{i,j,t}^{BM})$$

We can characterize our random coefficient risk-adjusting model in the following way:

$$R_{i,t}^{\kappa} = \alpha_i + \beta_i F_t + \varepsilon_{i,t}$$

where k = CAA, MT, SS. We assume that α_i and β_i are drawn independently from distributions with constant means and variances. F_t stands for year t factor returns. In order to risk-adjust the pension fund performance in all asset classes, we use the following factors: MKT (excess market return), SMB (small-minus-big), HML (high-minus-low), FIMKT (fixed income excess market return) and LIQ (traded liquidity factor). We add also MOM, (momentum factor) to the risk-adjusting model to control for returns on momentum trading strategies.

An important advantage of the random coefficient model is that it allows for heteroskedasticity and fund-specific betas, while being more robust to outliers than the standard Fama-MacBeth approach. As

 $^{^{20}}$ This return component, which measures the returns due to changes in strategic asset allocation policy over time, is different from the asset allocation component used in the decomposition of fund returns (Section 3). In this section, we look at the outcome of active decisions made by the pension fund to modify the strategic asset allocation policy in year *t* compared to year *t*-1. In the previous, we analyzed the return differences among funds, which can be attributed to the deviation in asset allocation policy from the average asset allocation policy of all funds in one year. Hence, the asset allocation component constructed to measure this deviation has a zero mean and cannot be used to measure return outcomes. The market timing and security selection components are constructed in the same way in sections 3 and 4.

Swamy (1970) explains, the random coefficient model is similar to a generalized least squares approach that puts less weight on the return series of funds that are more volatile.

In addition, we are also interested in the relation between certain pension fund characteristics and riskadjusted performance. Particularly, we would like to see whether characteristics like asset size, costs and investment style have a systematic impact on performance. These relations are tested using Fama-MacBeth (1973) regressions on risk-adjusted asset allocation, security selection and market timing components. In the first step, we perform a time-series regression on each fund's returns with an appropriate factor model. We run these regressions for every fund that has at least one more observation than coefficients to be estimated. Our findings do not change when we include only funds with at least 2, 3 or 4 more observations than coefficients (see Appendix Table A.2). Second, we run Fama-MacBeth regressions on alphas plus residuals retrieved in the risk-adjusting step, correcting standard errors for autocorrelation and heteroskedasticity using the Newey-West procedure.

For U.S. pension funds, we use MKT, SMB, HML, MOM from Kenneth French's website, plus fixed income and liquidity factors. To calculate fixed income excess returns (FIMKT), we use the returns on U.S. Broad Investment-Grade Bond Index (US BIG) from City Group.²¹ The traded liquidity factor is from Pastor and Stambaugh (2003). In the analysis of Canadian pension fund returns, we use the same factors, but calculated using the Canadian stock and fixed income markets. Stock excess returns (MKT) are calculated using the returns on the S&P TSX Index. SMB, HML and MOM factors for Canadian stock market were provided to us by Eun, Lai, de Roon and Zhang (2010). These factor loadings are calculated using the returns on more than 1,000 Canadian companies. We use the DEX Universe Bond Index, which is designed to be a broad measure of the Canadian investment-grade fixed income market, as a proxy for the Canadian fixed income market return.²²

4.2 Risk-adjusted Performance of Pension Funds

This subsection shows performance evaluation results of U.S. and Canadian pension funds. We start with analyzing performance on a fund level and then look separately at performance in equity, fixed income, real estate and private equity. Our focus is on the changes in asset allocation, market timing and security selection return components as explained previously.

4.2.1 Performance at the pension fund level

Table 6 provides summary statistics of benchmark-adjusted asset allocation, market timing and security selection returns, showing that U.S. and Canadian pension funds beat their benchmarks using

²¹ The US BIG Index is composed of the following securities: treasuries (excluding inflation-indexed securities and STRIPS); agencies (excluding callable zeros and bonds callable less than one year from issue date); mortgage pass-throughs; asset-backeds; supranationals; credit (excluding bonds callable less than one year from issue date); Yankees, globals, and securities issued under Rule 144A with registration rights.

²² In the DEX index there are four main credit or borrower categories: bonds issued by the Government of Canada (including Crown Corporations), Provincial bonds (including provincially guaranteed securities), Municipal Bonds, and Corporate Bonds.

all three components. For every variable we run a random coefficient regression with only a constant. Changes in the asset allocation policy produce 8 basis points (z-statistic of 2.20) return per year for U.S. funds, whereas Canadian funds exhibit a return of 4 basis points, which is not significant (z-statistic of 1.20). Market timing delivers about 28 basis points (z-statistic of 7.55) return per year to U.S. funds and 26 basis points (z-statistic of 7.10) to Canadian funds. Additionally, U.S. (Canadian) funds obtain 19 (28) basis points per year by security selection (z-statistics of 2.20 and 2.76, respectively).

We estimate a random coefficient model to assess whether the outperformance remains after riskadjusting. This is also important because benchmarks are generally chosen (and reported) by the funds themselves, such that funds could potentially choose benchmarks that are relatively easy to beat. The standard model we employ included five factors, namely the standard three Fama-French factors (market, size and value) augmented with the Pastor-Stambaugh (2003) traded liquidity factor and the excess return on a fixed income market index. We compare results using this baseline 5-factor model with using a 6-factor model that also includes a momentum factor. The momentum factor captures the difference in returns between a portfolio of stocks with high prior one-year returns (winners) and a portfolio of stock with low prior returns (losers). We do not consider the momentum factor as a priced risk factor, but rather include the momentum factor to understand its importance for the performance (attribution) of the funds in our sample. However, the question whether momentum is a priced risk factor (or can be explained by systematic risk) is clearly debatable, and it is straightforward to reinterpret our results under the assumption that momentum is priced (in which case funds would not get any credit on a risk-adjusted basis for successfully pursuing momentum strategies). Most of the literature, however, suggests that momentum cannot be explained by exposure to systematic risk factors (see e.g. Jegadeesh and Titman (1993, 2001), Lee and Swaminathan (2000), Cooper, Gutierrez and Hameed (2004) and Cremers and Pareek (2011)). Even papers arguing for a risk-based interpretation acknowledge that momentum cannot be mostly or completely explained by risk (see e.g. Grundy and Martin (2001) and Lu and Zhang (2008)).

Results in Table 7 show the annual alpha and beta coefficients on these factors, plus the root mean squared error (RMSE) of the residuals. After risk-adjusting, the changes in asset allocation policy deliver a positive alpha of 16 basis points per year (z-statistic of 2.72). This positive alpha is mainly due to the performance of U.S. funds, which obtain an asset allocation alpha of 22 basis points (z-statistic of 2.77), while the asset allocation alpha of Canadian funds is positive but insignificant. Inclusion of the momentum factor increases the estimated asset allocation alpha of U.S. funds by about 14 basis points to 35 basis points per year (z-statistic of 2.20). This suggests that changes in target weights are not made in order to capture momentum.

In Panel B, using all funds together, we find that after risk-adjusting market timing still delivers a positive alpha of 27 basis points per year (z-statistic of 5.68). For U.S. funds only, the results indicate even larger abnormal returns: 29 basis points (z-statistic of 5.08). Our results indicate that Canadian

funds deliver market timing abnormal returns of 24 basis points per year (z-statistic of 2.88).²³ Adding the momentum factor to the market-timing risk-adjustment model does not change our conclusion.

The beta coefficients indicate that pension funds, on average, do not systematically overweight a particular style. There is an economically small positive (but statistically significant) coefficient on the SMB factor that is driven by U.S. funds, but all other coefficients are statistically indistinguishable from zero. These results confirm the findings in Table 4, as the time averages of the mean differences for all asset classes are close to zero. However, Table 4 shows that pension funds' actual weights fluctuate substantially around reported policy weights. The results in Panel B of Table 7 show that these fluctuations covary positively with benchmark returns, evidenced by the positive coefficient of the constant, indicating market timing skill.²⁴

Panel C of Table 7 shows the random coefficient model results for security selection (i.e., benchmarkadjusted net returns). After risk-adjusting, security selection delivers a positive alpha of 45 basis points per year (z-statistic of 2.94), or 28 basis points for U.S. funds (z-statistic of 1.70, p-value of 0.091) and 83 basis points in our Canadian subsample. The positive security selection alpha of U.S. funds is fully driven by momentum. Once we add the momentum factor, U.S. funds' performance becomes negative: -107 basis points per year. This indicates that momentum trading strategies on average deliver around 135 basis points annually. We do not consider momentum as a priced risk factor, because most of the literature concludes that it cannot be captured by exposure to systematic risk factors. However, momentum is a well-known trading strategy and pension funds should seem to be able to implement it with relatively low investment costs. The large negative alpha after controlling for momentum returns indicates the failures of other active management strategies. Only part of this negative alpha can be attributed to investment costs.

The security selection abnormal return of Canadian funds is not sensitive to inclusion of the momentum factor. The alpha equals 89 basis points (z-statistic of 2.94), even when we control for momentum. However, when analyzing the performance of Canadian funds, it is critical to control for the so-called "Nortel effect". For this reason we include a year 2000 dummy variable. In this year, the Nortel company dominated the S&P/TSX Composite index. As of July 26 (2000), when Nortel was trading at \$124.50 a share on the TSX, it represented 36.5% of the TSE 300 index (http://www.cbc.ca/news/background/nortel/stock.html). The "Nortel effect" made it virtually impossible for a manager with a diversified Canadian equity portfolio to beat the benchmark index when Nortel's stock was outperforming the overall market. Following the index was also dangerous, because a portfolio with 36.5% invested in one company cannot qualify for diversified passive investing and is exposed to substantial idiosyncratic risk. However, later in the autumn of 2000, the

²³ When analyzing the performance of Canadian funds, we run a random coefficient model with and without the traded liquidity factor of Pastor and Stambaugh (2003), because this factor is based on U.S. stock market data. The inclusion of liquidity factor did not influence the estimated alphas in Tables 7 and 8. We used the traded liquidity factor to proxy for liquidity risk, which is considered as a highly relevant risk for large institutional investors like pension funds.

²⁴ Our results are robust to including only pension funds with a higher number of observations per fund in the regressions as presented by the robustness checks in Appendix Table A.2.

Nortel stock price started decreasing, dragging also the Canadian stock index down. The overall return of Nortel in 2000 was -33.8%, but until July that year Nortel had a positive return of 69%. The large negative returns on the Nortel stock created a substantial difference between the return on TSE Composite Index (7.4% in 2000) and the capped version of the same index (19.1% in 2000).

The abnormal returns of 89 basis points per year of Canadian funds disappear once we control for the "Nortel effect". After adjusting for 'Nortel effect' Canadian funds produce an alpha of -21 basis points (z-statistic of -0.97). This shows that Canadian funds beat significantly the S&P/TSX Composite index in 2000 due to the underweighting of Nortel in their portfolios (or good timing). The large impact of the "Nortel effect" on the overall fund performance can be expected, because Canadian pension funds' domestic equity holdings were 33% in 2000. As abnormal returns of Canadian funds disappear once we control for the year 2000, we can conclude that Canadian pension funds cannot consistently produce abnormal returns, but their "lucky" (or constrained) investment decisions around the "Nortel effect" resulted in returns significantly higher than the TSX benchmark returns.²⁵

Overall, our paper provides evidence that pension funds obtain a positive alpha of 17 basis points from changes in strategic asset allocation and 27 basis points from market timing asset allocation decisions. Security selection produces an even larger alpha of 45 basis points per year, which is robust to controlling for risks related to equity market, size, value, liquidity and fixed income market factors. For U.S. funds, any outperformance in security selection seems completely due to funds pursuing momentum strategies. For Canadian funds, the outperformance is due to the "Nortel" effect. The alpha for benchmark-adjusted net returns of all funds is -62 basis points per year, once we control for both momentum and the "Nortel effect" (z-statistic of -4.02). A large part of this negative return can be attributed to investment costs. The negative alpha from security selection among U.S. funds after controlling for momentum outweighs the positive risk-adjusted returns from changes in asset allocation and market timing. The negative, but insignificant, security selection alpha of Canadian funds after removing the "Nortel effect" also offsets most of the positive returns from market timing and asset allocation changes.

4.2.2 Performance in separate asset classes

Table 6 presents the summary statistics of pension funds asset allocation, market timing and security selection returns on an asset class level. Changes in the strategic asset allocation policy within equity produce small returns indistinguishable from zero. Before risk-adjusting, U.S. pension funds can beat their equity benchmarks by about 22 basis points (z-statistic of 5.98) per year using market timing and by an additional 19 basis points using security selection (z-statistic of 1.83 and p-value of 0.067). Results for Canadian funds are 17 and 43 basis points respectively. Canadian funds appear to be especially successful in their domestic equity investments: 64 basis points.

²⁵ Our results are robust to including only pension funds with a higher number of observations per fund in the regressions as presented by the robustness checks in Appendix Table A.2.

Within fixed income, U.S. funds seem to perform better than Canadian funds. Funds in both countries experience high variation in private equity and real estate performance, which follows from the substantially higher standard deviations, compared to equity and fixed income. As previously discussed, hedge funds become especially popular at the end of our period. Pension funds' returns in hedge funds are volatile and on average negative: -182 basis points (z-statistic of -1.85) in the U.S. and -227 basis points (z-statistic of -1.89) in Canada.²⁶ In line with our results, Dichev and Yu (2011) also find that the returns of hedge fund investors over 1980-2008 period are reliably lower that the returns of broad-based indexes like the S&P500 and only marginally higher than risk-free rates of return.

Risk-adjusted performance results for separate asset classes are presented in Table 8. In Panels A and B we show results for all funds, in Panels C and D for U.S. funds only and in Panels E and F for Canadian funds. The first column shows the results for changes in asset allocation within equity (AA-E), presenting alphas and root mean square errors of the random coefficient model using MKT, SMB, HML and LIQ factors. In Panels B, D and F we add the momentum factor to the risk-adjusting model.

Consistent with our findings in Table 6, changes in the asset allocation policy within equity do not deliver positive and significant alpha. Adding momentum does not influence the results. This suggests that our result (see Panel A of Table 7) that changes in asset allocation policy over time produced about 16 basis points alpha on the fund level, are not due to changes in the strategic allocation within equity, but are rather due to changes in policy weights across broader asset classes over time. For example, U.S. funds on average increased their policy allocation to private equity and other alternative assets at the expense of fixed income and cash.

The second column of Table 8 shows the market timing results within equity (MT-E), which can be measured only when funds invest in at least two categories within equity.²⁷ Results in the MT-E column confirm the findings in Table 7 Panel B that pension funds can create abnormal returns from timing their allocation decisions, obtaining around 14 basis points (z-statistic of 3.72). Controlling for the momentum factor, the alphas from market timing within equity are somewhat higher. As the market timing results at the fund level produced about 27 basis points alpha per year, this implies that a large part of this abnormal return is due to timing the performance across various equity classes, rather than timing the performance across broader asset classes (such as moving from equity to fixed income or real estate).

In line with our findings in Table 7, U.S. pension funds on average demonstrate security selection skills within equity (see column SS-E). The alpha from random coefficient regressions on equity

²⁶ These are the most frequently reported benchmarks by US and Canadian funds: U.S. equity – S&P500, Russell 1000, Russell 2000 and Russell 3000; Canadian equity – TSE300; U.S. fixed income – Citi Group US Big Index and Barclays US Aggregate; Canadian fixed income – DEX Universe Bond; Real estate – NCREIF, RCPI and NAREIT; Private equity – Wilshire 5000, Cambridge Private Equity, Venture Economics and custom benchmarks; Hedge funds – CSFB Tremont, HFRI Indices and custom benchmarks.

²⁷ For example, based on the strategic policy a pension fund should invest in 50% in Canadian equity, 30% in U.S. equity and 20% in EAFE equity. If the actual allocation percentages are different from the above-mentioned policy weights, that fund will generate returns from market timing within equity, measured as the difference between actual and policy weights times the benchmark returns. However, it does not capture returns from overweighting certain industries within Canadian equity mandate.

benchmark-adjusted net returns is equal to 38 basis points (z-statistic of 1.93). However, when we also control for the momentum factor, the alpha from security selection within equity turns negative to -72 basis points. Canadian funds exhibit a high security selection alpha before controlling for the "Nortel effect" (see Appendix Table A.3), but after controlling for the year 2000 and momentum, they obtain a large positive alpha in domestic equity of 90 basis points (z-statistic of 2.68). However, their overall return from security selection is still negative due to the poor performance in international equity.

Following Blake, Elton and Gruber (1993), Elton, Gruber and Blake (1995) and Cici and Gibson (2010), we risk-adjust the performance of fixed income assets using the following factors: MKT (equity market), FIMKT (fixed income market), HY (high yield) and OPTION (option-like characteristics of mortgage securities). Specifically, HY is the return difference of the Merrill Lynch High Yield and Government index for U.S. funds (or return difference between DEX BBB Universe and DEX Government Universe Index for Canadian funds). OPTION is estimated as the return difference of the US BIG (Canadian DEX) Mortgage Index and US BIG (Canadian DEX) Government Index.

Results in column AA-FI indicate that changes in the strategic allocation policy do not deliver abnormal returns for U.S. or Canadian funds. Further, column MT-FI presents the alphas obtained from market timing within fixed income assets. Results for all funds together as well as separately for U.S. and Canadian funds indicate that pension funds are not able to generate abnormal returns from timing their allocations between various fixed income categories. Moreover, security selection results in the column SS-FI show that U.S. and Canadian funds are not able to generate alphas from active management within fixed income assets. Alphas are also very close to zero when we look at the pension fund performance in domestic fixed income assets. Importantly, alphas disappear only after controlling for the high yield spread and option elements in fixed income returns. Our findings for pension fund performance are in line with Cici and Gibson (2010), who consider corporate-bond mutual funds.

Finally, we consider pension fund performance in private equity and real estate, the alternative asset classes with the highest number of observations. We again risk-adjust using market, size, value, and liquidity factors. The only adjustment is that we proxy for the equity market return using the excess returns on the Nasdaq index in the private equity regressions. This provides a better fit and thus is more representative for private equity investments.²⁸ Results in Table 8 indicate that private equity delivers a large positive alpha of 443 basis points per year (z-statistic of 3.07) on a risk-adjusted basis. After controlling for momentum, the alpha from private equity is considerable smaller and equal to 120 basis points per year, and no longer statistically different from zero. The results for real estate indicate a large negative alpha of about 1% a year, which is significant when combining U.S. and

²⁸ If we risk-adjust the security selection returns in private equity using the standard market return (CRSP data), we obtain significant alpha of 293 basis points (604 basis points without momentum). Table 8 shows that this alpha is lower (without controlling for momentum) or not present when we use the Nasdaq index as a proxy for excess market return.

Canadian funds. Real estate returns are not sensitive to adding a momentum factor to the risk-adjusting model.

4.3 Relation between Pension Fund Characteristics and Performance

In this section, we relate the risk-adjusted asset allocation, market timing and security selection alphas to certain characteristics of pension funds using Fama-MacBeth regressions. Specifically, we examine whether differences in performance are associated with fund size, mandate (or asset class) size, liquidity, investment costs and investment style (referring to whether assets are managed internally or externally, and passively or actively). Fund size reflects the total size of the pension fund holdings, which is a sum of holdings in all asset classes, while mandate size reflects the size of the holdings in a particular mandate, like fixed income. The analysis again is first conducted on a fund level and later by asset class.

4.3.1 Fund Level

Tables 9, 10 and 11 present the results for the asset allocation, market timing and the security selection component, respectively. Panel A of each table presents results when we risk-adjust using a five factor model that includes MKT, SMB, HML, LIQ and FIMKT in the first step of the Fama-MacBeth regressions. In Panel B thereof these tables, we add momentum to the five factor risk-adjusting model.

The asset allocation performance regressions with only a constant (Model 1 in Table 9) serve as a robustness check of the results in Panel A of Table 7. They confirm our previous finding that pension funds generate abnormal returns of 18 basis points per year from changes in the strategic asset allocation. Our results also confirm that this abnormal return is higher for U.S. funds, 25 basis points (t-statistic of 6.53), and lower for Canadian funds (8 basis points with a t-statistic of 2.29).

Fund size is slightly positively related to abnormal returns from changes in asset allocation policy. More importantly, the interaction between fund size and the fund's liquidity beta (i.e. the exposure to the Pastor-Stambaugh traded liquidity factor) is negative and highly significant. Using all funds, the interaction coefficient equals -1.68 (t-statistic of -18.04) in Model 3 of Table 10. Economically, this coefficient means that increasing liquidity by lowering the liquidity beta by 10 percentage points, would be associated with an improvement of the alpha of funds at the 75th size percentile by 19 basis points per year (= -0.1 x -1.6760 x (ln[5769]-ln[1896]) more than the improvement of the alpha of a fund at the median size percentile. This finding implies that larger funds face significant liquidity limitations when redesigning their strategic asset allocation. Rapid shifts in the strategic asset allocation towards more illiquid assets hurt the performance of larger funds relative to smaller funds.

Table 10 considers the relation between market timing returns and pension fund characteristics. The results with only a constant (Model 1 in Table 10) confirm our findings from Panel B of Table 7. They confirm our previous finding that pension funds generate abnormal returns of 28-31 basis points per year from market timing. Our results also confirm that this abnormal return is slightly higher for U.S.

funds, 30 basis points (t-statistic of 5.04), and lower, but still significant, for Canadian funds (25 basis points with a t-statistic of 5.20).

U.S. funds with higher costs are better at market timing. E.g., (in Panel B Model 4) the coefficient on total costs of 0.55 (t-statistic of 3.27) means that a one standard deviation increase in the total costs of U.S. funds (0.19) is associated with a 10 basis points per year higher return from market timing (0.19 * 0.55). In general, a positive coefficient on total costs implies that pension funds that obtain higher returns from market timing pay more for the flexibility to rebalance more frequently. Canadian funds do not show any association between market timing and costs. Therefore, the results are consistent with Canadian funds allowing for less flexibility to rebalance frequently, resulting in lower overall costs but lower market timing performance.

Fund size by itself does not seem related to the market timing abilities of pension funds. However, the interaction between fund size and the fund's liquidity beta is negative and highly significant. For the all funds universe the interaction coefficient equals -1.38 in Model 3 of Table 10. Economically, this coefficient means that increasing liquidity by lowering the liquidity beta by 10 percentage points, would be associated with an improvement of the alpha of funds at the 75th size percentile by 15 basis points per year (= -0.1 x -1.3778 x (ln[5769]-ln[1896]) more than the improvement of the alpha of a fund at the median size percentile.

Further, our results (in Appendix Table A.4) with the momentum factor included show that especially a high involvement in external management (%Ext) reduces market timing. External mandates usually include commitment periods and redemption fees, which make the rebalancing less attractive. Additional robustness checks confirm that size of the assets managed externally is negatively associated with market timing returns.²⁹

Overall, our results imply that smaller funds can rebalance their positions easily, without distorting market prices. Managing larger holdings externally reduces the market timing returns because of liquidity effects.

In Table 11, we consider the same fund characteristics for security selection (benchmark-adjusted net returns). Again, regressions with a constant only (Model 1) serve as a robustness check of alphas presented in Table 7, indicating again that U.S. funds on average display security selection skills, with average outperformance here estimated to be 49 basis points per year (t-statistic of 2.17; before adjusting for momentum, see Panel A). At the fund level, U.S. fund security selection performance is unrelated to fund size before controlling for momentum. However, if we control for momentum, we find that benchmark-adjusted returns become positively related to fund size. This suggests that it is

²⁹ In Appendix Tables A.4 and A.5 as a further robustness check we also look directly at the log of fund holdings invested in one style. For example SizeAct represents log of assets managed actively. This calculation of log styles' asset size is subject to arbitrary correction of the values when style holdings are equal to zero i.e. when no assets are managed in the particular style. We replace these results with ln(0.05), which will be the lowest value in every style. When we use these logs of style holdings in the regressions of Appendix Table A.4 our findings remain the same as in Table 10. In Tables 10 and 11 we do not include jointly in a regression the percentage and amount of assets managed with particular style due to high correlations. For example, the correlation between %Ext and SizeInt is -78% and the correlation between %Act and SizePas equals -69%.

especially smaller U.S. pension funds that are pursuing momentum strategies. As momentum can explain any evidence of outperformance of security selection for U.S., this could further explain the results in Bauer, Cremers and Frehen (2010) that small funds have particularly good equity performance.³⁰ In Canada, contrary to the U.S., larger funds follow more momentum strategies. Hence, economies of scale disappear once we control for momentum returns.

The security selection performance of larger funds seems particularly constrained by liquidity, as evidenced by the large, negative coefficients on the interaction between fund size and the liquidity beta. The interaction coefficient equals -1.3154 for the all funds universe in Model 3 of Table 11. Economically, this coefficient means that increasing liquidity by lowering the liquidity beta by 10 percentage points, would be associated with an improvement of the alpha of funds at the 75th size percentile by 15 basis points per year (= -0.1 x -1.3154 x (ln[5769]-ln[1896]) more than the improvement of the alpha of a fund at the median size percentile. Moreover, the magnitude of this interaction term decreases by half once we control for momentum (see Models 3 and 5). This shows that larger funds cannot engage as much as smaller funds in momentum trading due to liquidity constraints.

For Canadian funds, we consistently control for the "Nortel effect", which is especially important for smaller Canadian pension funds and thus matters greatly when considering the association of fund size with performance³¹. Once we control for the year 2000 outperformance of Canadian funds, the size effect becomes positive if we do not control for momentum. However, after controlling for momentum, size is no longer significant, suggesting that in Canada, larger funds are more actively involved in momentum strategies. If we look at Model 2 before controlling for momentum, one standard deviation increase in log size of Canadian funds improves their abnormal return by 10 basis points (0.0694 * 1.4686).

In Models 4 and 5, we further relate the total investment costs to security selection performance. For U.S. funds, before controlling for momentum, its coefficient is positive but insignificant (see Model 5 in Panel A; a coefficient of 0.49 with a t-statistic of 0.75), but it becomes negative and strongly significant after controlling for momentum (see Panel B; coefficient of -1.23 with a t-statistic of 2.45). This suggests that funds pursuing momentum strategies have higher costs that more than compensate with higher performance of these same strategies. Canadian funds show no significance for investment costs, and consistently with earlier results, adjusting for momentum does not matter.

In Model 6 of Table 11 and Appendix Table A.5, we analyze the association between security selection performance and the percentage of assets managed in active and external mandates. We first consider U.S. funds. In Model 6, the coefficient on the percentage of actively managed assets is

³⁰ These results are confirmed when forming three different samples based on the size of the assets under management. All three groups have, on average, positive security selection performance after risk-adjusting but before controlling for momentum (28 basis points per year for the smallest and 23 basis points for the largest third of U.S. pension funds). After controlling for momentum, all 3 groups have negative alpha, especially the smaller funds (-171 basis points per year for the smallest third and -92 basis points per year for the largest third of U.S. funds). Results are available upon request.

³¹ See Appendix Table A.6 for results without controlling for "Nortel effect"

positive but insignificant before adjusting for momentum, and becomes negative and significant (coefficient of -0.50 with a t-statistic of 1.68 or a p-value of 0.093) after adjusting for momentum. Likewise, the coefficient on the percentage of externally managed assets is only negative and significant after again adjusting for momentum. Both of these results can be explained by momentum being (obviously) an active strategy, and one that is apparently more pursued by external managers rather than by managers investing funds internally. Second, for Canadian funds, both coefficients are negative and generally statistically significant, indicating that funds with more passive and with more internal management perform better.

The result that active and external mandates are related to lower performance for U.S. funds (after controlling for momentum) can be explained by liquidity constraints faced by larger funds. Once we control for the interaction between size and the liquidity beta in Model 4 of Appendix Table A.5 Panel D, both become insignificant. This is consistent with liquidity constraints being most important for actively managed mandates and for assets managed externally. However, liquidity constraints fail to explain the negative association between active and external management for Canadian funds.³²

4.3.2 Asset Class Level

We also look at the influence of pension fund characteristics on performance on a lower level of aggregation, or how size, liquidity, costs and investment style relate to the performance in equity, fixed income, private equity and real estate. Table 12 shows the results for pension fund performance in equity, using the 4-factor equity model without momentum in Panel A and adding momentum in Panel B³³. We focus on the market timing and security selection returns within equity, as our previous findings showed that asset allocation returns due to changes in equity policy weights were indistinguishable from zero.

Results from Model 1 with a constant confirm the previous findings from Table 8. U.S. pension funds on average generate positive and significant alpha from both market timing across equities and security selection within equities. U.S. small cap performance is particularly large and positive (confirming Bauer, Cremers and Frehen (2010)). Canadian funds again perform especially well in security selection of domestic equities.

As equity represents on average 54%-58% of the total pension fund holdings, the results in Table 12 will naturally resemble the results in Tables 10 and 11 at the fund level. For equities, size is strongly negatively related to performance for both U.S. and Canadian funds (see Model 2 in Panel A). This is largely related to liquidity constraints (see Model 3 in Panel A). For U.S. funds, the pursuit of momentum strategies explains (see Panel B) the direct effect of asset size, but not of the liquidity-size interaction. Cost levels are actually positively related to the security selection of U.S. funds for their domestic equity and small cap equity investments (see Model 4 in Panel A), which can again be

³² When we use these logs of style holdings in the regressions of Appendix Table A.5 our findings remain the same as in Table 11. Greater internal management is related to improved performance.

³³ Appendix Table A.8 presents the results for U.S. and Canadian funds together and the results without controlling for "Nortel effect".

explained by momentum (see Model 4 in Panel B). Much of the higher costs of small cap equity investing can thus be explained by funds' pursuit of momentum, which seems an investment strategy bearing higher investment costs than other active strategies.

In line with our findings on the fund level, after controlling for momentum, the percentage of equity holdings managed internally is positively associated with security selection and market timing in equity, especially among U.S. funds (see Model 5 in Panel B). One standard deviation increase in the percentage of equity holdings managed externally by U.S. pension fund reduces their security selection performance by 49 basis points (0.2435 * (-2.0167)). The only exception is the performance of U.S. funds in small cap mandates. We still observe a positive relationship, which is due to the better performance of smaller funds in small cap mandates. One explanation is that smaller funds engage less in internal management and another potential explanation is that investing in small cap stocks requires more expertise that cannot be easily internally acquired.

Further, there is no significant relationship between the percentage of actively managed equity holdings and the performance of U.S. pension funds. Among Canadian funds, the percentage of active management is negatively related to performance, which implies that managing equity more passively reduces their underperformance in equity (see Model 5 in Panel B). One standard deviation increase in the percentage of equity assets managed actively by Canadian funds increases the underperformance by 18 basis points (0.2221 * (-0.8217)).

Pension fund performance in fixed income assets is analyzed in Table 13.³⁴ In line with Table 8, pension funds do not obtain positive alphas from market timing or security selection within fixed income. After risk-adjusting for the 4-factor fixed income model, their alphas are close to zero and not statistically significant (see Model 1). The size of the fixed income holdings leads to differences in performance among U.S. funds (See Model 2 in Panel A). Similar to equity results, smaller funds display better security selection skills, but the economic magnitude is smaller. A one standard deviation increase in the log of fixed income holdings leads to an 18 basis points (1.51 * (-0.1168)) decrease in risk-adjusted returns from security selection. Somewhat surprising, investment costs are positively related to performance in fixed income (see Model 3). This result is mainly driven by better performance of smaller mandates, which bear higher costs, and is also economically relatively small. A one standard deviation increase in fixed income investment costs of U.S. funds improves their performance by nine basis points (0.11 * 0.7888). The investment style variables %ActFI and %ExtFI also have a marginal impact on security selection and market timing returns, but the economic effect is less than 10 basis points for a one standard deviation change. Compared to equity, the influence of pension fund characteristics has a much lower effect on the fixed income performance.

Finally, in Table 14 we look at the performance in the two most important alternative asset classes: real estate and private equity. In the first step, risk-adjusting is done using the 4-factor model with Fama-French factors augmented with liquidity. In the private equity analysis, we use the excess returns of the Nasdaq index as a proxy for the market return. In Panels B and D, we add the

³⁴ Appendix Table A.9 shows the results for U.S. and Canadian funds together.

momentum factor to the model. Model 1 in Panels A and B confirms our findings in Table 8 that U.S. and Canadian funds generate significant and positive alphas in private equity before controlling for momentum. Afterwards, the alpha is insignificant and close to zero. Results in Model 1 also indicate that pension funds tend to have negative alphas from real estate.

In Panel B, the results also show that funds with larger holdings perform better in private equity. This effect is strongest among Canadian pension funds. A one standard deviation increase in the log of private equity holdings of Canadian pension funds leads to 600 basis points (2.6341 * 2.3) higher abnormal return (see Model 3 in Panel B). Similarly, Canadian funds with large holdings in real estate also realize economies of scale. A one standard deviation increase in the log of their real estate holdings leads to 158 basis points (0.8294 * 1.9) higher abnormal return (see Model 3 in Panel D). The economic magnitude of the other coefficients is also large. The performance of larger funds in alternative assets is again constrained by liquidity, as evidenced by the large, negative coefficients on the interaction between mandate size and the liquidity beta (see Model 4).

Our results show that larger funds can assert more bargaining power in private deals in private equity, infrastructure and real estate.³⁵ Larger funds can devote more resources to monitor closer their external counterparts in these investment vehicles. The largest funds even establish internal (or at-arms-length") private equity and real estate divisions, which is a good long-term approach if you are large enough. The negotiation power of funds that invest large amounts in private equity or real estate can be seen in the coefficients on investment costs (see Models 5 and 6). Costs are negatively related to abnormal returns from private equity and real estate and the effects are especially pronounced among U.S. funds investing in private equity and Canadian pension funds investing in real estate.

These economies of scale in private equity and real estate investments (as well as the negative relation between costs and performance) are not driven by funds that recently started investing in these asset classes. Controlling for 'new investors' in these asset classes with less experience and possibly worse opportunities, the coefficients on size, costs and liquidity interaction term remain unchanged (results are not reported). Our results are also robust to not risk-adjusting in the first step, but rather directly regressing the net benchmark-adjusted private equity and real estate returns on characteristics.

Overall, the previous discussion shows that the relationship between size and performance is not uniform. Large funds experience diseconomies of scale in more standardized asset classes, like domestic equity and fixed income, but manage to recover by economies of scale in alternative asset classes.

³⁵ Lopez-de-Silanes, Phalippou and Gottschalg (2010) identify strong diseconomies of scale among private equity firms. Our findings do not contradict the observed diseconomies of scale in the private equity industry. First, pension funds holdings in private equity can be compared to the value of investments held in parallel by private equity firms. The median holding of pension funds would fall in the second smallest decile in Lopez-de Silanes et al. (2010) sorting of private equity firms, based on value of investment held in parallel. Second, pension funds have an option to invest in multiple private equity companies if they are concerned about diseconomies of scale. Third, we look at returns on investments in private equity after subtracting costs, and private equity is the most expensive asset class in pension fund portfolio.

5. Persistence in Pension Fund Performance

Previous sections showed that pension funds obtain positive returns from market timing and security selection, some of which remain significant even after risk-adjusting. A relevant follow-up question is whether there is persistence in pension fund performance. To answer this question we split pension funds into five quintiles based on their market timing on security selection performance (after costs). We run an ordered logit model, where the dependent variable is the quintile ranking based on the performance in year t+1 and the main independent variable is the quintile ranking in year t. Marginal effects from the ordered logit model for every outcome (quintile ranking) are presented in Table 15. Results for the U.S. are in Panels A and B, and for Canada in Panels C and D. Market timing performance is considered in Panels A and C, and security selection in Panels B and D.

Results indicate that funds are more likely to end up in a better performing quintile next year, if they belong to a better performing quintile already this year. Pension funds are also more likely to be ranked among the worst performers next year, if they performed relatively poorly this year. The persistence is observed among both U.S. and Canadian funds and in both market timing and security selection returns. For example, looking at U.S. funds market timing returns (Panel A), an increase in the quintile ranking from 3 to 4 reduces the probability of ranking among the worst performers in year t+1 by 3.8%. Results in Models 2, 3 and 4 show that the marginal effects of last year's ranking remain even after controlling for fund size, costs, and the percentage of assets managed actively and externally.

In Appendix Table A.11 we present the actual transition matrixes. The percentage of funds repeating as best performers is in all cases higher than the percentage of best performers of last year ending in one of the four lower quintiles this year. The same holds for the worst performing funds. We also look at the returns in year t+1 of funds ranked in the lowest and highest quintile in year t. Funds ranked in the top quintile have higher average returns in the following year than the funds ranked in the bottom quintile. Results from the t-test suggest that the difference in next year's returns is significant for U.S. funds' market timing and security selection returns and for Canadian funds' security selection returns.

These persistence tests are performed directly on the benchmark-adjusted market timing and security selection returns. As we only have access to annual data, we cannot use the risk-adjusted performance in these estimations. Hence, we do not test whether pension funds can persistently deliver abnormal returns, or estimate the effect of liquidity constraints on persistence. Nevertheless, these results show that certain pension funds are persistently better in outperforming their benchmarks using market timing and security selection.

6. Conclusion

We analyze the asset allocation policy and performance of U.S. and Canadian defined benefit pension funds. We provide a detailed overview of the time trends in pension fund allocations to multiple asset classes and the investment costs of pension funds. There is a substantial home bias in pension fund allocations to equity and fixed income that goes down over time. Real estate is the most important alternative asset class for pension funds and accounts for 3-4% of their total holdings, whereas private equity is by far the most expensive asset category with mean annual expenses of 280 basis points.

We decompose differences between pension fund returns into asset allocation policy, market timing and security selection returns. We find that pension funds manage approximately 80% of their total assets in an active way, which creates substantial differences in their returns. Even though pension funds have the opportunity to invest in multiple asset classes, the majority of them follow a standard asset allocation policy. This 'herding' in asset allocation together with high levels of active management explains why security selection accounts for more than 50% of the variation in excess returns of both U.S. and Canadian pension funds.

Pension funds are able to beat their benchmarks before and after risk-adjusting. Changes in asset allocation policy result in positive abnormal returns of 17 basis points per year, which is mainly observed among U.S. funds. These abnormal returns are due to pension funds changing their asset allocation policy across broader asset classes over time, not to changes within equity or fixed income. This suggests that when modifying their strategic asset allocations, especially larger funds face significant liquidity limitations.

Market timing delivers a positive alpha of 27 basis points per year. This abnormal return is larger among smaller funds, funds with greater internal management and funds with higher investment costs. In general, market timing returns depend on the flexibility to rebalance and liquidity constraints. About half of the alpha comes from market timing within different equity styles (such as domestic versus international stocks, and large versus small cap stocks). This suggests that funds that try to stay as close as possible their strategic asset allocation policy may miss market timing opportunities. If fund managers have market timing skills, as our results indicate, letting the actual weights deviate from the strategic weights and not rebalancing back immediately can in fact improve performance, in line with Sharpe's (2010) idea of an 'adaptive asset allocation policy.'

Security selection delivers an even higher alpha of 45 basis points, where the outperformance among U.S. funds is driven by the momentum factor and that of Canadian funds can be fully explained by the "Nortel effect" in the year 2000. Once we control for these two factors, security selection delivers a negative alpha of -62 basis points per year. For U.S. funds, we find that costs have a strong negative effect on performance after controlling for momentum. This implies that the momentum strategy bears higher costs than other active strategies. Further, we find that the security selection performance of larger funds is particularly constrained by liquidity.

The relation between size and performance is not uniform and depends on the asset class and investment style. Larger pension funds experience diseconomies of scale in equity and fixed income (mainly due to liquidity limitations), but they realize their economies of scale in alternative asset classes, especially in real estate. Larger funds can assert more negotiation power in alternative asset classes, which enables them to access better investment opportunities at lower costs. Our results also show that funds that manage most of their assets internally improve their performance compared to

peers with mostly external mandates, potentially due to fewer agency conflicts and lower investment costs.

Lastly, we find persistence in pension funds' ability to deliver higher market timing and security selection returns. Funds belonging to the best performing quintile this year are more likely to remain among the best performers in the following year.

References:

Allen, F. 2001. Presidential Address: Do Financial Institutions Matter?, *Journal of Finance*, 56(4), p.1165-1175.

Bauer, R., M.K.J. Cremers, and R. Frehen. 2010. Pension Fund Performance and Costs: Small Is Beautiful, *Working Paper*.

Bikker, J.A., O.W. Steenbeek and F. Torracchi. 2010. The Impact of Scale, Complexity and Service Quality on the Administrative Costs of Pension Funds: a Cross Country Comparison, *Working Paper*.

Blake, C.R., E. Elton, and M.J. Gruber. 1993. The Performance of Bond Mutual Funds, *Journal of Business*, 66(3), p.371-403.

Blake, D., B.N. Lehmann, and A. Timmermann. 1999. Asset Allocation Dynamics and Pension Fund Performance, *Journal of Business*, 72(4), p.429-461.

Blake, D., A. Timmermann, I. Tonks, and R. Wermers. 2010. Decentralized Investment Management: Evidence from the Pension Fund Industry, *Working Paper*.

Brown, K.C., L. Garlappi, and C. Tiu. 2010. Asset Allocation and Portfolio performance: Evidence from University Endowment Funds, *Journal of Financial Markets*, 13, p.268-294.

Carhart, M.M. 1997. On persistence in Mutual Fund Performance, Journal of Finance, 52(1), p.57-82.

Chan, L.K., H.L. Chen, and J. Lakonishok, 2002, On mutual fund investment styles, *The Review of Financial Studies*, 15(5), p.1407-1437.

Chen, J., H. Hong, M. Huang, and J.D. Kubik. 2004. Does Fund Size Erode Mutual Fund Performance? The role of Liquidity and Organization, *American Economic Review*, 94(5), p.1276-1302.

Cici, G., and S. Gibson. 2010. The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings, *Working Paper*.

Cooper, M.J., R.C. Gutierrez Jr. and A. Hameed. 2004. Market States and Momentum, *Journal of Finance*, 59(3), p1345-1365.

Cremers, K.J.M., M. Ferreira, P. Matos, and L. Starks. 2011. The Mutual Fund Industry Worldwide: Explicit and Closet Indexing, Fees, and Performance, *Working Paper*.

Cremers, K.J.M. and A. Pareek. 2011. Short-Term Trading and Stock Return Anomalies: Momentum, Reversal, Accruals, Share Issuance and R&D Increases, *Working Paper*.

Cremers, K.J.M., and A. Petajisto. 2009. How Active Is Your Fund Manager? A New Measure That Predicts Performance, *The Review of Financial Studies*, 22(9), p.3329-3365

Dichev, I.D., and G. Yu. 2011. Higher Risk, Lower Returns: What Hedge Fund Investors Really earn, *Journal of Financial Economics*, 100(2), p.248-263.

Dyck, A. and L. Pomorski. 2011. Is Bigger Better? Size and Performance in Pension Plan Management, *Working Paper*.

Elton, E.J., M.J. Gruber, and C.R. Blake. 1995. Fundamental Economic Variables, Expected Returns, and Bond Fund Performance, *Journal of Finance*, 50(4), p.1229-1256.

Eun, C.S., S. Lai, F.A. de Roon, and Z. Zhang. 2010. International Diversification with Factor Funds, *Management Science*, 56(9), p.1500-1518.

Fama, E. F., and J. D. MacBeth. 1973. Risk, Return, and Equilibrium: Empirical Tests, *Journal of Political Economy*, 81(3), p.607-636.

French, K.R. 2008. Presidential Address: The Cost of Active Investing, Journal of Finance, 63(4), p.1537-1573.

Fung, W., D.A. Hsieh, N.Y. Naik, and T. Ramadorai. 2008. Hedge Funds: Performance, Risk, and Capital Formation, *Journal of Finance*, 63(4), p.1777-1803.

Goyal, A., and S. Wahal. 2008. The Selection and Termination of Investment Management Firms by Plan Sponsors, *Journal of Finance*, 63(4), p.1805-1847.

Gruber, M.J. 1996. Another Puzzle: The Growth in Actively Managed Mutual Funds, *Journal of Finance*, 51(3), pp.783-810.

Grundy, B.D. and J.S. Martin. 2001. Understanding the nature of the risks and the source of the rewards to momentum investing, *Review of Financial Studies*, 14(1), p.29-78.

Ibbotson, R.G. 2010. The Importance of Asset Allocation, Financial Analysts Journal, 66(2), p.18-20

Ibbotson, R.G. and P.D. Kaplan. 2000. Does asset Allocation Policy Explain 40, 90 or 100 Percent of Performance?, *Financial Analysts Journal*, 56(1) p.26-33

Jegadeesh, N., and S. Titman. 1993. Returns to buying winners and selling losers: implications for stock market efficiency, *Journal of Finance*, 48, p.65-91.

Jegadeesh, N., and S. Titman. 2001. Profitability of Momentum Strategies: An Evaluation of Alternative Explanations, *Journal of Finance*, 56, p.699-720.

Kacperczyk, M., C. Sialm, and L. Zheng. 2008. Unobserved Actions of Mutual Funds, *Review of Financial Studies*, 21(6), p.2379-2416

Kosowski, R., A. Timmermann, R. Wermers, and H. White. 2006. Can Mutual Fund "Stars" Really Pick Stocks? New Evidence from a Bootstrap Analysis, *Journal of Finance*, 61(6), p.2551-2594

Lakonishok, J., A. Shleifer, and R.W. Vishny. 1992. The structure and performance of the money management industry. *Brookings Papers on Economic Activity, Microeconomics*, p.339-391.

Lee, C., and B. Swaminathan. 2000. Price momentum and trading volume, *Journal of Finance*, 55, p.2017-2069.

Liu, L.X. and L. Zhang. 2008. Momentum Profits, Factor Pricing, and Macroeconomic Risk, *Review of Financial Studies*, 21(6), p.2417-2448.

Lopez-de-Silanes F., L. Phalippou and O. Gottschalg. 2009. Giants at the Gate: Diseconomies of Scale in Private Equity, *Working Paper*.

Lucas, D. and S.P. Zeldes. 2009. How should public plans invest? *American Economic Review*, 99(2), p.527-532.

Malkiel, B.G. 1995. Returns from Investing in Equity Mutual Funds 1971 to 1991, *Journal of Finance*, 50(2), p.549-72.

Novy-Marx R. and J. Rauh. 2011. Public Pension Promises: How Big Are They and What Are They Worth? *Forthcoming in Journal of Finance*.

Pastor, L. and R.F. Stambaugh. 2003. Liquidity Risk and Expected Stock Returns, *Journal of Political Economy*, 111(3), p.642-685.

Phalippou, L. 2009. Beware of Venturing into Private Equity, *Journal of Economic Perspectives*, 23(1), p.147-166.

Sirri, E.R. and P. Tufano. 1998. Costly Search and Mutual Fund Flows, *Journal of Finance*, 53(5), p.1589-1622.

Sharpe, W.F. 2010. Adaptive Asset Allocation Policies, Financial Analysts Journal, 66(3), p.45-59.

Swamy, P.A.V.B. 1970. Efficient Inference in a Random Coefficient Regression Model, *Econometrica*, 38(2), p.311-323.

Timmermann, A., and D. Blake. 2005. International Asset Allocation with Time-Varying Investment Opportunities, *Journal of Business*, 78(1), p.71-97.

Xiong, J.X., R.G. Ibbotson, T.M. Idzorek, and P. Chen. 2010. The Equal Importance of Asset Allocation and Active Management, *Financial Analysts Journal*, 66(2), p.22-30.

Table 1: Number of funds

This table displays the number of U.S. and Canadian pension funds in the CEM database in the sample period 1990-2008. We also display the number of funds entering and exiting the database in a given year. The Total row shows the total number of funds reporting at least one year to the CEM.

Veen		Total			U.S.			Canada	
rear	# Funds	# Enter	# Exit	# Funds	# Enter	# Exit	# Funds	# Enter	# Exit
1990	104	104	0	51	51	0	53	53	0
1991	124	58	38	63	39	27	61	19	11
1992	164	69	29	83	38	18	81	31	11
1993	220	90	34	134	70	19	86	20	15
1994	266	92	46	168	68	34	98	24	12
1995	294	83	55	192	62	38	102	21	17
1996	290	57	61	185	36	43	105	21	18
1997	265	37	62	168	29	46	97	8	16
1998	278	47	34	174	37	31	104	10	3
1999	292	56	42	182	40	32	110	16	10
2000	270	38	60	165	23	40	105	15	20
2001	276	46	40	177	36	24	99	10	16
2002	254	30	52	156	15	36	98	15	16
2003	254	39	39	158	27	25	96	12	14
2004	262	43	35	166	25	17	96	18	18
2005	262	36	36	155	15	26	107	21	10
2006	247	31	46	146	18	27	101	13	19
2007	311	102	38	215	88	19	96	14	19
2008	290	46	67	204	36	47	86	10	20
Total	774			534			240		

Table 2: Cox proportional hazard model and self-reporting bias

This table presents the results of a survival analysis using the Cox proportional hazard model. The event of interest is the decision of the pension funds not to report to CEM in a given year. We treat each fund re-entry as a new fund which explains why the # Units is higher than the # Funds presented in Table 1. # Exit Events presents the number of observations when pension funds decided not to report to CEM again. # Obs. presents the total number of observations in the database. Independent variables included in the model are Net returns in percentage points, Benchmark-adjusted net returns in percentage points, Total Costs in basis points and Log(Size) – logarithm of the asset under management. In this table the hazard ratios for each independent variable are reported together with their corresponding z-statistics in parentheses. Panel A presents the results for all funds, while in Panels B and C we split the sample on U.S. and Canadian funds. All regressions use robust standard errors clustered by year.

Interpretation of the hazard ratios in Panel A:

Net returns: 1 percentage point increase in the Net returns increases the rate of dropping out from the CEM Database by 2.22%. Total costs: when the total costs increase by 1 basis point, the dropping rate decreases by (100% - 99.5%) = 0.5%. Log(Size): when the log(size) increase by 1 unit, the dropping rate decreases by (100% - 74.83%) = 25.17%.

# Units	# Exit	# Obs.	Net returns	Benchmark	Benchmark-	Total Costs	Log(Size)
	Events			returns	adj net returns		
Panel A: All Fi	unds						
1078	787	4643	1.0222			0.9950	0.7483
			(1.81)			(-1.03)	(-6.81)
1078	787	4643			1.0060	0.9953	0.7549
					(0.44)	(-1.04)	(-6.89)
1078	787	4643		1.0274	1.0049	0.9951	0.7470
				(1.70)	(0.44)	(-1.04)	(-6.85)
Panel B: U.S.							
736	532	2910	1.0220			0.9915	0.7339
			(1.86)			(-1.76)	(-7.50)
736	532	2910			1.0120	0.9924	0.7514
					(0.78)	(-1.60)	(-5.50)
736	532	2910		1.0274	1.0076	0.9915	0.7294
				(1.70)	(0.54)	(-1.79)	(-7.91)
Panel C: Cana	ıda						
342	255	1733	1.0209			0.9925	0.6670
			(1.20)			(-1.31)	(-7.63)
342	255	1733			0.9790	0.9907	0.6588
					(-0.93)	(-1.41)	(-7.62)
342	255	1733		1.0245	0.9904	0.9924	0.6702
				(1.30)	(-0.38)	(-1.32)	(-7.44)

Table 3: Fund size, asset allocation and costs summary statistics

Panel A presents the size and percentage allocation summary statistics. Funds Size is reported in million USD for U.S. funds and in million CAD for Canadian funds. Domestic E presents the percentage allocation to domestic equity from total holdings, whereas Domestic FI displays the percentage allocation to domestic fixed income from total holdings. Other category incorporates the remaining alternative asset classes: tactical asset allocation, infrastructure, hedge funds, commodities and natural resources. Panel B presents the costs summary statistics. Total costs row presents the total fund costs in basis points. The costs are also reported separately for equity, domestic equity, fixed income, domestic fixed income, cash, real estate, private equity and other alternative assets (tactical asset allocation, infrastructure, hedge funds, commodities and natural resources). For every variable the time series averages of cross-sectional means, 25th percentile and 75th percentile values for the period 1990–2008 are presented.

		U.S.			Canada	
	25 th perc.	Mean	75 th perc.	25 th perc.	Mean	75 th perc.
Panel A: Asset un	der Managen	ient (Holding	gs) summary ste	atistics		
Fund Size	1234.55	9559.66	8109.81	496.55	4347.84	2557.39
Equity	52.68%	58.42%	65.60%	49.81%	54.04%	59.10%
Domestic E	38.46%	44.84%	51.78%	25.50%	30.49%	35.83%
Fixed Income	24.03%	31.08%	36.51%	32.70%	37.64%	42.32%
Domestic FI	19.96%	26.99%	33.46%	27.09%	32.50%	40.20%
Cash	0.21%	1.94%	2.56%	0.70%	3.44%	4.76%
Real Estate	0.34%	3.97%	6.36%	0.19%	2.94%	4.93%
Private Equity	0.00%	2.38%	3.87%	0.00%	0.77%	0.73%
Other	0.00%	2.21%	1.15%	0.00%	1.17%	0.30%
Panel B: Costs su	mmary statist	ics in basis p	points			
Total Costs	23.04	35.25	46.03	17.06	25.65	30.92
Equity	20.06	32.92	44.39	19.55	29.18	35.39
Domestic E	16.19	29.41	40.74	13.71	22.37	27.77
Fixed Income	10.86	19.30	25.67	7.22	14.08	19.85
Domestic FI	9.38	17.99	24.41	6.87	13.82	20.02
Cash	3.48	28.27	17.82	4.39	13.35	15.94
Real Estate	53.97	89.14	106.35	24.27	54.03	74.52
Private Equity	138.25	283.81	292.84	73.53	273.93	313.32
Other	36.16	95.27	113.38	64.49	105.37	138.75

Table 4: Strategic policy weights vs. Actual weights

This table presents the strategic policy weights of the pension funds and the realized policy weights. Column Policy weight presents the time series averages of cross-sectional mean strategic policy weights (target weights) for different asset classes for the period 1990–2008. Column Actual weight presents the time series averages of cross-sectional mean realized weights for different asset classes for the period 1990–2008. Mean column of Actual_t – Policy_t displays the time series averages of cross-sectional mean differences between the actual weights and strategic policy weights, whereas the StDev column presents the time series average of cross-sectional standard deviations of the mean differences between the actual (realized) weights and strategic (target) weights. Mean column of Policy_t – Policy_{t-1} displays the time series averages of cross-sectional mean differences between the strategic policy weights in year *t* and the strategic policy weights in the previous year *t*-1, whereas the StDev column presents the time series average of cross-sectional standard deviations of the differences between the strategic policy weights from year *t* and year *t*-1.

	Policy weight	Actual woight	Actual _t -	– Policy _t	Policy _t –	Policy _{t-1}
	roncy weight	Actual weight	Mean	StDev	Mean	StDev
Panel A: U.S.						
Equity	58.30%	58.42%	0.11%	5.45%	0.02%	4.50%
Fixed Income	31.44%	31.08%	-0.36%	4.96%	-0.33%	4.36%
Cash	1.19%	1.94%	0.75%	2.32%	-0.13%	1.39%
Real estate	4.62%	3.97%	-0.65%	2.05%	-0.06%	1.62%
Private Equity	2.91%	2.38%	-0.52%	2.26%	0.22%	1.66%
Other	1.54%	2.21%	0.66%	3.67%	0.27%	2.25%
Panel B: Canada						
Equity	53.86%	54.04%	0.18%	4.74%	0.49%	3.45%
Fixed Income	38.66%	37.64%	-1.02%	4.56%	-0.29%	3.78%
Cash	2.61%	3.44%	0.82%	3.11%	-0.26%	1.74%
Real estate	3.31%	2.94%	-0.37%	1.96%	-0.01%	1.52%
Private Equity	0.70%	0.77%	0.07%	1.42%	0.02%	0.98%
Other	0.86%	1.17%	0.31%	2.48%	0.05%	1.64%

Table 5: Time-series and cross-sectional return variation

This table reports the time-series and cross-sectional summary statistics on a fund level, incorporating all assets. Panel A displays the summary statistics from the cross-sectional distribution of R-squared statistics obtained from performing the following regression for each pension fund over time:

$$R_{i,t} - PR_t = \alpha_i + b_i R_{i,t}^k + \varepsilon_{i,t} , \quad i = 1, \dots, n$$

where $R_{i,t}$ is the net return of pension fund *i* at time *t* and PR_t is the average of equally weighted policy return for year *t*. $R_{i,t}^k$ refers to asset allocation return component $R_{i,t}^{AA}$, the market timing component $R_{i,t}^{MT}$ and the security selection component $R_{i,t}^{SS}$. The three different return components are defined in section 3. Panel B reports the summary statistics from the time-series distribution of R-squared statistics from the 19 (1990–2008) cross-sectional regressions:

$$R_{i,t} - PR_t = \alpha_i + b_i R_{i,t}^k + \varepsilon_{i,t}, \quad t = 1, \dots, T$$

where k = AA, MT, SS. At least five data points per fund are required to run each time-series or cross-sectional regression. Hence, in the regressions on all funds we include 348 funds, of which 217 are U.S. funds and 131 Canadian funds. Appendix Table A.1 presents the results for other cutoff points – at least 4, 7 or 9 observations per fund.

	All Funds U.S.										
	Mean	Median	SD		Mean	Median	SD	Ν	/lean	Median	SD
Panel A: Time-series R-squared values											
Asset Allocation policy	59.63	66.30	28.60		41.78	38.51	29.76		33.76	31.20	28.22
Market Timing	17.82	9.45	21.06		18.35	9.73	21.78	1	7.62	9.31	22.34
Security Selection	33.59	28.60	25.95		54.69	56.36	28.06	4	58.20	65.62	28.63
Panel B: Cross-sectional	l R-squar	ed values									
Asset Allocation policy	51.13	51.54	22.41		35.07	35.49	18.32	4	24.72	21.15	14.69
Market Timing	4.52	2.01	5.11		4.06	3.50	3.66		7.22	4.19	10.65
Security Selection	21.88		44.62	43.26	19.82	2	18.25	46.48	19.98		

Table 6: Return components summary statistics

This table presents the asset allocation, market timing and security selection return components summary statistics. For every variable we run a random coefficient model only with a constant. In the Mean column the value of this constant is shown, while z-stat column presents the corresponding z-statistics. AA All Assets, MT All Assets and SS All Assets rows present the asset allocation, market timing and security selection returns on a fund level. We also report these three return components separately for equity (AA Equity, MT Equity and SS Equity) and fixed income investments (AA Fixed Income, MT Fixed Income and SS Fixed Income). For equity and fixed income assets the summary statistics for domestic investments are also included (SS Domestic Equity and SS Domestic Fixed Income). For private equity and real estate we present only the security selection returns (SS Private Equity and SS Real Estate). SS Hedge Funds summary statistics are based on shorter time period (2000-2008). The number of funds considered for these summary statistics corresponds to the number of funds included in the regressions in Tables 7 and 8.

	U.S. fur	ids only	Canadian Funds only
	Mean	z-stat	Mean z-stat
AA All Assets	0.0839	(2.20)	0.0429 (1.20)
MT All Assets	0.2770	(7.55)	0.2640 (7.10)
SS All Assets	0.1913	(2.20)	0.2846 (2.76)
AA Equity	-0.0362	(-1.16)	-0.0564 (-1.18)
MT Equity	0.2209	(5.98)	0.1700 (4.92)
SS Equity	0.1856	(1.83)	0.4272 (2.70)
SS Domestic Equity	0.0184	(0.16)	0.6415 (2.97)
AA Fixed Income	0.0217	(0.87)	0.0028 (0.17)
MT Fixed Income	0.0211	(0.83)	-0.0098 (-0.58)
SS Fixed Income	0.0952	(1.16)	-0.1060 (-1.59)
SS Domestic Fixed Income	0.1343	(1.55)	-0.1092 (-1.48)
SS Private Equity	0.5715	(0.57)	-1.1263 (-0.65)
SS Real Estate	-0.3201	(-0.98)	0.3417 (0.70)
SS Hedge Funds (2000-2008)	-1.8248	(-1.85)	-2.2731 (-1.89)

Table 7: Risk-adjusted performance per return components in all asset classes on a fund level

This table reports the net risk-adjusted performance at the fund level for all assets using a random coefficients model. The following factors are included in the regressions: MKT, SMB and HML are the Fama-French factor returns, MOM – momentum factor, LIQ – Pastor and Stambaugh (2003) traded liquidity factor, FIMKT – fixed income excess market return. Dummy Y2000 is a dummy variable for year 2000 (Nortel effect). In Panel A the dependent variable is the return due to changes in asset allocation policy, which is calculated as the return due to changes in the strategic asset allocation weights in year *t* compared to year *t*-1 multiplied with the benchmark return $R_{i,t}^{CAA} = \sum_{j=1}^{N} (w_{i,j,t}^{AA} - w_{i,j,t}^{AA}) r_{i,j,t}^{BM}$. In Panel B the dependent variable is the market timing component of fund returns $R_{i,t}^{MT} = \sum_{j=1}^{N} (w_{i,j,t} - w_{i,j,t}^{AA}) r_{i,j,t}^{BM}$ is the policy weight for fund *i* for asset class *j* and year *t*, $w_{i,j,t}$ is the actual realized weight for fund *i* for asset class *j* and year *t* and $r_{i,j,t}^{BM}$ is the benchmark return for fund *i* for the asset class *j* and period *t*. In Panel C we use the security selection component of fund returns as dependent variable $R_{i,t}^{SS} = \sum_{j=1}^{N} w_{i,j,t} (r_{i,j,t} - r_{i,j,t}^{BM})$, where $r_{i,j,t}$ is the realized net return on the asset class *j* for the year *t* by fund *i*. We report the annual alpha (Cons.) and betas with corresponding z-statistics in parentheses. RMSE is the root mean square error.

	# Funds # Obs	Cons.	MKT	SMB	HML	MOM	FIMKT	LIQ	Dummy X2000	RMSE
Panel A · Ch	anges in As	set Allocati	on return c	omponent	(Year t weig	ohts – Year	t-1 weights) * Renchm	ark returns	
All Funds	203	0 1662	-0.0078	0.0020	-0.0087	nus rear	0.0096	-0.0028		12 0981
7 m r unus	2585	(2, 72)	(-3.06)	(0.78)	(-3.55)		(1.04)	(-0.79)		12.0701
All Funds	203	0.2402	-0.0088	0.0007	-0.0106	-0.0026	0.0100	-0.0050		12 1071
i in i unus	2585	(2, 27)	(-2.45)	(0.27)	(-3.06)	(-0.77)	(1.16)	(-1 34)		12.1071
US	120	0.2181	-0.0091	0.0045	-0.0103	(0.77)	0.0123	-0.0064		13 3716
0.5.	1492	(2.77)	(-2.41)	(1.11)	(-3.00)		(0.92)	(-1.51)		10.0710
U.S.	120	0.3468	-0.0120	0.0027	-0.0145	-0.0061	0.0150	-0.0083		13,3903
	1492	(2.20)	(-2.12)	(0.67)	(-2.68)	(-1.10)	(1.20)	(-1.75)		
Canada	83	0.0787	-0.0057	-0.0008	-0.0070	(0.0090	0.0028		10.1814
	1093	(0.81)	(-1.93)	(-0.29)	(-2.08)		(0.76)	(0.46)		
Canada	83	0.0986	-0.0042	-0.0012	-0.0057	0.0015	0.0047	-0.0001		10.1773
	1093	(1.11)	(-1.34)	(-0.46)	(-1.75)	(0.55)	(0.41)	(-0.02)		
	1 . 77					• 7 •) **				
Panel B: Ma	irket Timing	return con	nponent: (A	ctual weigh	$\frac{\text{nts} - Policy}{0.0047}$	weights) * I	Benchmark	returns		10.0000
All Funds	256	0.2666	-0.0045	0.0059	-0.0047		0.0061	0.0010		12.0239
	3341	(5.68)	(-1.52)	(2.66)	(-1.92)	0.0005	(0.87)	(0.21)		10.0001
All Funds	256	0.2/4/	-0.0044	0.0046	-0.0045	-0.0005	0.0052	0.0006		12.0321
ЦС	3341	(4.14)	(-1.33)	(2.05)	(-1.54)	(-0.18)	(0.64)	(0.13)		12 2750
0.5.	152	0.2867	-0.0051	(2.45)	-0.0050		(1.47)	-0.0013		13.2/56
ЦС	1949	(5.08)	(-1.59)	(2.45)	(-1.80)	0.0046	(1.47)	(-0.35)		12 20 42
0.5.	152	0.3254	-0.0066	(1, (7))	-0.0058	-0.0046	(1.25)	0.0001		13.2943
Conside	1949	(3.22)	(-1.62)	(1.07)	(-1.05)	(-1.08)	(1.35)	(0.03)		10.0650
Canada	104	(2.83)	-0.0052	(1, 22)	-0.0035		-0.0024	(0.45)		10.0038
Canada	1392	(2.00)	(-0.37)	(1.55)	(-1.10)	0.0026	(-0.21)	(0.43)		10.0505
Canada	104	(3, 23)	-0.0011	(1.12)	-0.0058	(1.30)	-0.0078	(0.0001)		10.0393
	1392	(3.23)	(-0.19)	(1.12)	(-0.73)	(1.50)	(-0.70)	(0.01)		
Panel C: See	curity Select	tion return	component:	Actual wei	ghts * (Rea	lized net re	turn – Benc	hmark retu	rn)	
All Funds	253	0.4461	0.0089	0.0121	0.0338		-0.0675	-0.0179		11.9592
	3276	(2.94)	(1.65)	(2.11)	(5.08)		(-3.70)	(-2.14)		
All Funds	253	-0.2711	0.0237	0.0382	0.0598	0.0581	-0.1202	-0.0198		11.9115
	3276	(-1.35)	(3.42)	(4.98)	(8.22)	(5.22)	(-5.66)	(-1.92)		
All Funds	223	-0.0549	0.0081	0.0190	0.0121		-0.0380	0.0091	2.9992	12.0064
	3036	(-0.48)	(1.83)	(3.59)	(2.14)		(-2.48)	(1.38)	(9.06)	
All Funds	223	-0.6249	0.0225	0.0374	0.0403	0.0516	-0.0867	-0.0026	2.5755	11.9470
	3036	(-4.02)	(4.17)	(6.20)	(5.89)	(6.77)	(-5.34)	(-0.37)	(7.89)	
U.S.	152	0.2837	0.0085	0.0303	0.0219		-0.0819	-0.0086		13.1562
	1937	(1.70)	(1.18)	(3.43)	(2.71)		(-3.46)	(-1.01)		
U.S.	152	-1.0677	0.0369	0.0768	0.0653	0.1056	-0.1643	-0.0153		13.1211
	1937	(-4.45)	(3.71)	(6.79)	(6.84)	(6.29)	(-5.74)	(-1.16)		
Canada	101	0.8327	0.0077	-0.0120	0.0532		-0.0609	-0.0359		10.2089
	1339	(2.98)	(0.94)	(-2.66)	(4.87)		(-2.10)	(-2.21)		
Canada	101	0.8939	0.0053	-0.0128	0.0554	-0.0053	-0.0617	-0.0293		10.2672
	1339	(2.94)	(0.64)	(-2.58)	(4.90)	(-0.71)	(-2.01)	(-1.75)		
Canada	88	-0.0441	0.0004	-0.0008	0.0338		-0.0024	0.0108	3.2943	10.2724
	1235	(-0.23)	(0.05)	(-0.19)	(4.16)		(-0.11)	(0.88)	(6.71)	
Canada	88	-0.2103	0.0055	-0.0010	0.0397	0.0205	-0.0283	-0.0037	3.8044	10.2994
	1235	(-0.97)	(0.77)	(-0.22)	(4.72)	(2.66)	(-1.26)	(-0.29)	(7.46)	

Table 8: Risk adjusted returns per asset class

This table reports the net risk-adjusted performance at the asset class level using a random coefficients model. The asset allocation (AA), market timing (MT) and security selection (SS) return components within Equity are risk-adjusted using the following factors: MKT, SMB and HML – the Fama-French factor returns, LIQ – Pastor and Stambaugh (2003) traded liquidity factor and year dummy 2000 for all funds and Canadian funds (Panels A, B, E and F). In Panels B, D and F we also add MOM – momentum factor to the model. Domestic equity SS column displays the security selection returns (benchmark adjusted net returns) of U.S. Funds mandates in U.S. equity and Canadian Funds mandates in Canadian equity. Within fixed income asset allocation (AA-FI), market timing (MT-FI) and security selection (SS-FI) return components are risk-adjusted using the following factors: FIMKT – fixed income excess return, HY – high yield spread, OPTION – option-like characteristics of mortgage securities returns and MKT – equity excess return. Domestic fixed income SS-DFI column displays the security selection (benchmark adjusted net returns) for U.S. Funds mandates in U.S. fixed income and Canadian Funds mandates in Canadian fixed income. Private equity security selection returns (benchmark-adjusted returns) are regressed on excess returns of NASDAQ index and SMB, HML and LIQ factors. Real Estate security selection returns (benchmark-adjusted returns) are regressed on MKT, SMB, HML and LIQ factors. In panels B, D, and F we also add MOM – momentum factor to the private equity and real estate regressions. The table shows the alpha, its corresponding z-statistic and the root mean square error (RMSE) from all regressions.

AA: (Policy weight year t - Policy weight year t-1) * Benchmark returns

MT: (Actual weights - Policy weights) * Benchmark returns

SS: Actual weights * (Realized net return – Benchmark return)

		Equity		Domestic		Fixed Income	<u>,</u>	Domestic Fixed Inc	Private Fauity	Real Estate
	AA-F	MT-F	SS-F	SS-DF	AA-FI	MT-FI	SS-FI	SS-DFI	SS-PF	SS-RF
Panel A:	All Funds wi	thout adjust	ing for mome	ntum	71111		5511	55 011	SOIL	55 112
# Funds	230	256	252	249	261	232	318	298	122	186
# Obs.	2765	3287	3225	3188	2960	2814	3662	3346	1396	2165
Alpha	-0.0350	0.1435	-0.0351	0.1212	0.0167	-0.0030	-0.0296	0.0063	4.4332	-0.8219
1	(-0.70)	(3.72)	(-0.27)	(0.78)	(0.59)	(-0.11)	(-0.35)	(0.07)	(3.07)	(-2.07)
RMSE	12.1816	11.9577	12.3004	12.6749	12.0843	12.3910	11.7636	11.6572	16.3668	11.7609
Panel B:	All Funds wi	th adjusting	for momentu	um						
# Funds	230	256	252	249					122	186
# Obs.	2765	3287	3225	3188					1396	2165
Alpha	-0.0278	0.2139	-0.8027	-0.2522					1.2005	-0.9999
	(-0.20)	(3.79)	(-4.52)	(-1.25)					(0.55)	(-1.92)
RMSE	12.1982	11.9629	12.2779	12.6499					16.6544	11.7929
Panel C:	U.S. without	adjusting fo	r momentum							
# Funds	138	176	176	175	157	160	200	186	93	123
# Obs.	1609	2095	2078	2060	1732	1861	2231	2069	1067	1418
Alpha	-0.0692	0.1328	0.3839	0.1193	0.0368	0.0045	-0.0248	0.0258	4.7015	-0.7838
	(-1.22)	(3.05)	(1.93)	(0.48)	(0.92)	(0.12)	(-0.24)	(0.25)	(2.77)	(-1.67)
RMSE	13.3890	13.1782	13.6732	13.7768	13.2500	13.3155	12.5052	12.2935	16.3342	(12.5887)
Panel D:	U.S. with ad	justing for m	omentum							
# Funds	138	176	176	175					93	123
# Obs.	1609	2095	2078	2060					1067	1418
Alpha	-0.0695	0.2278	-0.7208	-0.5358					0.3555	-1.1112
	(-0.32)	(3.27)	(-2.70)	(-1.91)					(0.18)	(-1.62)
RMSE	13.4234	13.2064	13.5800	13.7611					17.0062	12.7525
Panel E:	Canada with	out adjusting	g for momen	tum						
# Funds	92	104	104	99	104	72	118	112	29	63
# Obs.	1156	1360	1343	1303	1228	953	1431	1277	329	747
Alpha	0.0286	0.1076	0.0146	1.0421	-0.0127	-0.0184	-0.0776	-0.0680	3.0452	-1.1128
	(0.31)	(1.77)	(0.06)	(3.52)	(-0.32)	(-0.72)	(-0.53)	(-0.42)	(1.10)	(-1.50)
RMSE	10.2902	10.0612	10.4133	11.3180	10.1723	10.2641	10.2490	10.2432	16.9253	9.7527
Panel F:	Canada with	adjusting fo	or momentum	!						
# Funds	92	104	104	99					29	63
# Obs.	1156	1360	1343	1303					329	747
Alpha	0.0387	0.1817	-0.5360	0.8997					2.1098	-0.9872
	(0.31)	(1.89)	(-2.07)	(2.68)					(0.31)	(-1.25)
RMSE	10.2933	10.0654	10.4291	11.3896					17.3202	9.6920

Table 9: Pension fund characteristics and total asset allocation (AA) returns

In the first step we regress the total asset allocation returns on a five factor model that includes the MKT, SMB, HML, LIQ and FIMKT. In Panels B and D we also add the momentum factor to the five factor model. We run these regressions for every fund that has at least 8 observations, which results in 203 Funds (2585 observations) in All funds models, 120 U.S. Funds (1492 observations) and 83 Canadian funds (1093 observations). In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). We include the following characteristics: LogSize – log of average pension fund holdings in a given year, %Act – percentage of all holdings invested in active mandates, %Ext – percentage of all holdings invested in external mandates and Costs – total fund costs. SizeLiq is an interaction term of the log fund size with the first step fund-specific loading on the liquidity factor. In parentheses we report the t-statistics for every coefficient.

	Model 1	Model 2		Model 3			Model 4		Model 5				Model 6		
	Cons	Cons	LogSize	Cons	LogSize	SizeLiq	Cons	Costs	Cons	LogSize	SizeLiq	Costs	Cons	%Act	%Ext
Panel A: wi	thout adjus	ting for mon	nentum in th	ne first step											
All Funds	0.1761	-0.3986	0.0704	0.1445	-0.0022	-1.6760	0.1026	0.2219	0.0503	0.0023	-1.6745	0.1849	0.4762	-0.4263	0.0634
	(7.27)	(-3.16)	(4.75)	(2.15)	(-0.37)	(-18.04)	(2.57)	(1.20)	(0.78)	(0.51)	(-18.53)	(1.02)	(5.86)	(-7.28)	(0.65)
U.S.	0.2461	-0.4773	0.0834	0.1693	-0.0049	-1.6679	0.1924	0.1406	-0.0919	0.0122	-1.6818	0.3204	0.2363	0.0895	-0.0593
	(6.53)	(-1.94)	(3.00)	(0.92)	(-0.27)	(-13.68)	(4.34)	(0.63)	(-0.80)	(1.06)	(-14.69)	(1.72)	(2.74)	(0.67)	(-0.37)
Can	0.0807	-0.0392	0.0168	0.1440	-0.0022	-1.5619	0.0552	0.1050	0.0499	0.0086	-1.5904	0.1362	0.9857	-1.0882	0.0122
	(2.29)	(-0.38)	(1.16)	(2.10)	(-0.24)	(-18.46)	(0.99)	(0.47)	(0.35)	(0.59)	(-20.76)	(0.48)	(6.28)	(-6.84)	(0.13)
Panel B: wi	th adjusting	g for momen	tum in the fi	irst step											
All Funds	0.2699	0.0426	0.0273	0.6274	-0.0587	-2.0494	0.1490	0.3980	0.5532	-0.0548	-2.0515	0.1514	0.6433	-0.6950	0.2278
	(11.37)	(0.33)	(1.81)	(4.30)	(-3.13)	(-7.45)	(3.26)	(3.23)	(2.20)	(-2.20)	(-7.40)	(0.67)	(7.36)	(-7.84)	(5.81)
U.S.	0.4112	0.6751	-0.0323	1.3945	-0.1490	-2.9698	0.3724	0.0807	1.6763	-0.1699	-2.9912	-0.3859	0.4732	-0.3227	0.2179
	(10.56)	(1.91)	(-0.84)	(4.32)	(-3.73)	(-9.52)	(4.49)	(0.39)	(3.05)	(-3.10)	(-9.89)	(-1.09)	(3.89)	(-2.09)	(3.08)
Can	0.0766	0.0878	-0.0012	0.1466	-0.0098	-0.3621	-0.0022	0.3654	-0.0233	0.0052	-0.3808	0.3220	1.0052	-1.1353	0.0319
	(2.31)	(0.93)	(-0.10)	(1.40)	(-0.72)	(-2.16)	(-0.05)	(1.55)	(-0.18)	(0.35)	(-2.33)	(1.49)	(5.80)	(-6.29)	(0.42)

Table 10: Pension fund characteristics and total market timing (MT) returns

In the first step we regress the total market timing returns on a five factor model that includes the MKT, SMB, HML, LIQ and FIMKT. In Panels B and D we also add the momentum factor to the five factor model. We run these regressions for every fund that has at least 8 observations, which results in 256 Funds (3297 observations) in All funds models, 152 U.S. Funds (1937 observations) and 104 Canadian funds (1360 observations). In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). We include the following characteristics: LogSize – log of average pension fund holdings in a given year, %Act – percentage of all holdings invested in active mandates, %Ext – percentage of all holdings invested in external mandates and Costs – total fund costs. SizeLiq is an interaction term of the log fund size with the first step fund-specific loading on the liquidity factor. In parentheses we report the t-statistics for every coefficient.

	Model 1	Model 2		Model 3			Model 4		Model 5				Model 6		
	Cons	Cons	LogSize	Cons	LogSize	SizeLiq	Cons	Costs	Cons	LogSize	SizeLiq	Costs	Cons	%Act	%Ext
Panel A: wi	thout adjust	ting for mom	entum in th	e first step											
All Funds	0.2757	0.3300	-0.0072	0.4403	-0.0193	-1.2403	0.1691	0.3731	0.3488	-0.0141	-1.2555	0.1774	0.1932	0.1094	-0.0025
	(5.67)	(5.18)	(-0.77)	(6.43)	(-2.12)	(-11.74)	(2.67)	(3.52)	(3.26)	(-1.75)	(-12.08)	(1.04)	(1.80)	(0.95)	(-0.03)
U.S.	0.3030	0.7421	-0.0532	0.6380	-0.0423	-1.3776	0.0766	0.7193	0.4030	-0.0255	-1.3658	0.3009	-0.0260	0.2240	0.1901
	(5.04)	(5.19)	(-3.00)	(4.65)	(-2.62)	(-10.35)	(1.19)	(5.11)	(4.59)	(-2.32)	(-10.16)	(1.97)	(-0.37)	(2.96)	(2.11)
Can	0.2534	0.1262	0.0177	0.2354	0.0083	-1.5065	0.3769	-0.5356	0.6362	-0.0284	-1.5447	-0.7229	0.3331	0.1017	-0.1963
	(5.20)	(1.23)	(1.79)	(2.14)	(0.68)	(-14.73)	(3.04)	(-1.56)	(1.26)	(-0.62)	(-16.78)	(-1.01)	(2.07)	(0.61)	(-2.82)
Panel B: wi	th adjusting	for moment	tum in the fi	rst step											
All Funds	0.3051	0.2789	0.0030	0.2126	0.0110	-1.3778	0.2107	0.3580	-0.0607	0.0289	-1.3897	0.5230	0.4578	-0.0817	-0.0940
	(6.22)	(3.11)	(0.33)	(2.02)	(1.13)	(-7.20)	(5.53)	(2.82)	(-1.03)	(4.53)	(-7.27)	(3.00)	(4.13)	(-0.67)	(-0.93)
U.S.	0.3500	0.4177	-0.0091	0.4126	-0.0012	-2.0574	0.1798	0.5493	0.0712	0.0266	-2.0510	0.4056	0.4957	-0.0473	-0.1193
	(4.92)	(1.87)	(-0.45)	(1.52)	(-0.05)	(-8.06)	(3.08)	(3.27)	(0.56)	(1.84)	(-7.72)	(1.67)	(3.96)	(-0.33)	(-1.16)
Can	0.2268	0.5345	-0.0422	0.7466	-0.0782	-1.2438	0.2983	-0.2907	1.0401	-0.1114	-1.3149	-0.3923	0.2260	0.1414	-0.1380
	(4.91)	(3.40)	(-2.17)	(5.27)	(-4.52)	(-14.86)	(2.74)	(-1.16)	(1.93)	(-2.10)	(-12.53)	(-0.57)	(1.75)	(1.37)	(-1.14)

Table 11: Pension fund characteristics and security selection (SS) returns

In the first step we regress the total security selection returns on a five factor model that includes the MKT, SMB, HML, LIQ and FIMKT. In Panels B and D we also add the momentum factor to the five factor model. The regressions for all funds and Canada contain also year dummy 2000. We run these regressions for every fund that has at least 8 observations (9 observations if we include year dummy 2000 in the first step), which results in 224 Funds (3044 observations) in All funds models, 152 U.S. Funds (1937 observations) and 88 Canadian funds (1235 observations). In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). We include the following characteristics: LogSize – log of average pension fund holdings in a given year, %Act – percentage of all holdings invested in active mandates, %Ext – percentage invested in external mandates and Costs – total fund costs. SizeLiq is an interaction term of the log fund size with the first step fund-specific loading on the liquidity factor. In parentheses we report the t-statistics for every coefficient.

	Model 1	Model 2		Model 3			Model 4		Model 5				Model 6		
	Cons	Cons	LogSize	Cons	LogSize	SizeLiq	Cons	Costs	Cons	LogSize	SizeLiq	Costs	Cons	%Act	%Ext
Panel A: wi	ithout adjust	ting for mon	nentum in th	he first step											
All Funds	0.0446	-0.4409	0.0619	-0.4692	0.0780	-1.3154	-0.0549	0.3720	-0.4216	0.0722	-1.3159	-0.0365	0.5325	-0.2068	-0.3753
	(0.37)	(-2.27)	(2.72)	(-2.54)	(3.59)	(-11.90)	(-0.55)	(2.11)	(-1.73)	(2.80)	(-11.11)	(-0.16)	(2.91)	(-0.74)	(-2.65)
U.S.	0.4897	0.5503	-0.0075	0.1931	0.0150	-1.5839	0.2362	0.8779	-0.1352	0.0351	-1.5357	0.4853	0.2124	0.0493	0.3085
	(2.17)	(2.26)	(-0.54)	(1.37)	(0.79)	(-7.13)	(1.75)	(1.43)	(-0.24)	(0.67)	(-6.57)	(0.75)	(0.94)	(0.14)	(0.91)
Can	-0.0806	-0.5814	0.0694	-0.7227	0.1119	-1.1716	-0.0154	-0.2387	-0.3578	0.0940	-1.2051	-0.8926	1.0618	-0.5560	-0.8258
	(-0.69)	(-3.08)	(4.66)	(-4.85)	(6.99)	(-19.65)	(-0.16)	(-0.38)	(-0.78)	(1.94)	(-20.13)	(-0.93)	(6.77)	(-3.72)	(-4.56)
Panel B: wi	ith adjusting	for momen	tum in the f	irst step											
All Funds	-0.6290	-0.3396	-0.0359	-0.2648	-0.0429	-0.7724	-0.3862	-0.8768	0.5026	-0.0929	-0.8389	-1.4371	0.0092	0.0055	-0.7727
	(-8.01)	(-1.55)	(-1.54)	(-1.05)	(-1.69)	(-4.31)	(-5.80)	(-3.32)	(1.87)	(-3.39)	(-4.98)	(-3.07)	(0.03)	(0.02)	(-4.44)
U.S.	-0.9907	-2.8922	0.2266	-2.5915	0.1841	-0.6865	-0.4687	-1.5846	-1.4949	0.1009	-0.6787	-1.2332	0.1265	-0.4975	-0.8184
	(-6.67)	(-8.47)	(7.03)	(-5.70)	(3.86)	(-2.38)	(-3.52)	(-3.97)	(-2.51)	(1.95)	(-2.30)	(-2.45)	(0.33)	(-1.68)	(-3.09)
Can	-0.3191	-0.5704	0.0319	-0.5519	0.0332	-0.6600	-0.3954	0.5661	-0.9398	0.0852	-0.7190	0.5725	0.5566	-0.3273	-0.7064
	(-3.68)	(-1.63)	(0.72)	(-1.66)	(0.78)	(-9.79)	(-1.83)	(0.42)	(-0.98)	(1.00)	(-5.84)	(0.25)	(2.52)	(-1.56)	(-2.36)

Table 12: Equity – pension fund characteristics and performance

In the first step we regress the equity security selection (SS) (net benchmark-adjusted returns) or market timing (MT) return component on a four factor model that includes the MKT, SMB, HML and LIQ. We run these regressions for every fund that has at least 7 observations. For Canadian funds we also add year dummy 2000 to the factor model and run regressions for every fund with at least 8 observation in that case. In Panel B we also add MOM – momentum factor to the model. In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). The following characteristics are included in the Fama-MacBeth regressions: LogMand – log of the total equity holdings, Costs – costs for investing in equity, %ActE – percentage in active mandates and %ExtE – percentage in external mandates from the equity holdings. For U.S. Small Cap %ActE and %ExtE are estimated based on assets in U.S. small cap equity. Mand_Liq is an interaction term of the log mandate size with the first step fund-specific loading on the liquidity factor. The first column # Funds and # Obs. present the number of funds and the number of observations included in the analysis. In parentheses we report the t-statistics for every coefficient.

	# Funds	Model 1	Model 2		Model 3			Model 4		Model 5		
	# Obs.	Cons.	Cons.	LogMand	Cons	LogMand	Mand_Liq	Cons	Costs	Cons	%ActE	%ExtE
Panel A: without adjusting	for moment	um in the first	step									
U.S. Equity SS	176	0.5654	2.4441	-0.2371	0.0817	0.0133	-1.7477	0.5699	0.1340	-0.7303	2.2447	-0.3921
	2078	(2.35)	(4.93)	(-5.63)	(0.12)	(0.15)	(-9.99)	(1.30)	(0.09)	(-1.08)	(1.50)	(-0.65)
U.S. Equity MT	175	0.1915	0.2064	-0.0009	0.3743	-0.0228	-0.8289	0.2452	-0.1736	0.2423	0.1115	-0.1540
	2068	(3.07)	(0.92)	(-0.04)	(1.66)	(-1.01)	(-6.81)	(2.74)	(-0.91)	(4.47)	(0.96)	(-3.08)
U.S. Domestic Equity	175	0.4858	4.2150	-0.4812	1.8326	-0.2074	-1.5540	0.0453	1.7296	-0.4851	1.3535	0.0103
SS	2060	(2.07)	(2.90)	(-2.95)	(1.27)	(-1.30)	(-8.13)	(0.25)	(2.75)	(-1.92)	(2.37)	(0.04)
U.S. Domestic Equity	86	1.3725	5.7826	-0.5618	4.7260	-0.4694	-1.5829	0.3697	2.3301	-0.8160	0.8983	1.6867
Small Cap SS	857	(2.51)	(3.19)	(-3.11)	(2.53)	(-2.42)	(-7.28)	(0.66)	(2.67)	(-1.06)	(1.81)	(2.50)
Canada Equity SS	104	0.0071	0.5491	-0.0876	0.7146	-0.1111	-1.0443	0.3872	-1.3147	0.9023	-0.4663	-0.5470
	1343	(0.04)	(1.58)	(-2.16)	(2.10)	(-2.89)	(-9.13)	(2.04)	(-1.88)	(2.03)	(-1.37)	(-1.56)
Canada Equity MT	111	0.1218	0.3455	-0.0351	0.4060	-0.0408	-1.1187	0.1337	-0.0598	0.0605	0.0337	0.0432
	1392	(2.08)	(2.74)	(-2.87)	(3.66)	(-2.87)	(-3.45)	(4.36)	(-0.32)	(1.12)	(0.21)	(0.32)
Canada Domestic Equity	99	0.9736	1.7593	-0.1404	1.5886	-0.1026	-1.5368	1.1965	-0.8869	0.9888	0.2840	-0.2820
SS	1303	(4.93)	(4.07)	(-1.81)	(4.18)	(-1.78)	(-11.75)	(3.15)	(-0.58)	(1.79)	(1.58)	(-0.64)
Panel B: with adjusting for	r momentum	in the first ste	D									
U.S. Equity SS	176	-0.7573	-0.5634	-0.0230	-2.4983	0.1918	-1.6017	0.1148	-2.8050	-0.1141	1.5036	-2.0167
	2078	(-4.80)	(-1.01)	(-0.39)	(-1.76)	(1.07)	(-5.26)	(0.26)	(-1.92)	(-0.13)	(0.83)	(-2.62)
U.S. Equity MT	175	0.3063	0.4538	-0.0168	0.6680	-0.0444	-0.8693	0.3250	-0.0595	0.4704	0.0104	-0.1963
	2068	(5.12)	(1.13)	(-0.38)	(1.51)	(-0.89)	(-4.11)	(3.06)	(-0.28)	(4.95)	(0.08)	(-3.73)
U.S. Domestic Equity	175	-0.5537	0.8706	-0.1779	-0.0969	-0.0730	-1.3649	-0.0588	-1.8839	0.0830	0.9044	-1.5932
SS	2060	(-2.95)	(0.56)	(-1.00)	(-0.07)	(-0.50)	(-6.20)	(-0.30)	(-4.88)	(0.19)	(0.89)	(-3.92)
U.S. Domestic Equity	86	-0.6947	5.4173	-0.7898	4.2499	-0.7432	-2.5245	-1.0779	1.0083	-2.1444	0.3825	1.3608
Small Cap SS	857	(-1.36)	(2.42)	(-3.13)	(2.11)	(-3.34)	(-8.43)	(-1.83)	(1.09)	(-2.96)	(0.66)	(2.23)
Canada Equity SS	104	-0.6444	0.2164	-0.1365	0.3217	-0.1423	-1.2704	-0.4268	-0.5952	0.2755	-0.8217	-0.2271
	1343	(-4.17)	(0.65)	(-4.07)	(1.15)	(-10.86)	(-2.43)	(-1.76)	(-0.44)	(0.83)	(-2.21)	(-0.85)
Canada Equity MT	111	0.1896	0.5538	-0.0575	0.5494	-0.0574	0.0481	0.1325	0.2176	-0.0662	0.2897	0.0323
	1392	(4.71)	(5.17)	(-4.63)	(5.10)	(-4.92)	(0.41)	(3.82)	(1.88)	(-0.82)	(2.29)	(0.28)
Canada Domestic Equity	99	0.8132	1.8665	-0.1831	1.2671	-0.0167	-1.9786	0.8556	0.2598	0.5801	0.2739	0.0216
SS	1303	(4.10)	(7.29)	(-4.21)	(2.15)	(-0.14)	(-2.94)	(1.71)	(0.12)	(0.91)	(0.90)	(0.06)

Table 13: Fixed income – pension fund characteristics and performance

In the first step we regress the fixed income security selection (SS) (net benchmark-adjusted returns) or market timing (MT) return component on a four factor model that includes the FIMKT, MKT, OPTION and HY. We run these regressions for every fund that has at least 6 observations. In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). The following characteristics are included in the Fama-MacBeth regressions: LogMand – logarithm of the total fixed income holdings, Costs – costs for investing in Fixed Income, %ActFI – percentage in active mandates from the fixed income holdings and %ExtFI – percentage in external mandates from the fixed income holdings. The first column # Funds and # Obs. present the number of funds (cross-sectional units) and the number of observations included in the analysis. Panel A presents the results for U.S. funds only, while Panel B shows the results for Canadian funds. In parentheses we report the t-statistics for every coefficient. Market timing component within fixed income requires that the fund invests in at least two types of fixed income (for example: Canadian fixed income and EAFE fixed income). In that case there can be a difference in weights within the fixed income, which will lead to return component that is due to the difference from actual and strategic weights.

		Model 1	Model 2		Model 3		Model 4		
	# Funds	Cons.	Cons.	LogMand	Cons	Costs	Cons	%ActFI	%ExtFI
	# Obs.								
Panel A: U.S.									
FI SS	200	0.0389	0.8504	-0.1168	-0.0856	0.7888	-0.1895	0.1169	0.1719
	2231	(0.33)	(1.95)	(-2.25)	(-0.66)	(1.78)	(-0.98)	(0.41)	(1.69)
FI MT	160	0.0299	-0.1094	0.0186	0.0270	0.0251	-0.0186	0.1262	-0.0721
	1861	(1.93)	(-1.49)	(1.75)	(0.85)	(0.23)	(-0.58)	(2.37)	(-2.27)
Domestic FI SS	186	-0.0350	0.2183	-0.0374	-0.1792	0.9794	0.1109	-0.0582	-0.1025
	2069	(-0.31)	(1.41)	(-1.51)	(-1.70)	(2.37)	(0.59)	(-0.22)	(-1.15)
Panel B: Canada									
FI SS	118	-0.0073	0.1310	-0.0203	-0.2122	1.7752	-0.2157	0.2031	0.0633
	1430	(-0.09)	(0.59)	(-0.60)	(-2.29)	(1.34)	(-0.95)	(0.88)	(0.46)
FI MT	72	-0.0149	0.0044	-0.0028	-0.0449	0.2404	-0.0082	-0.0387	0.0391
	953	(-1.59)	(0.09)	(-0.44)	(-1.94)	(1.09)	(-0.32)	(-1.24)	(1.80)
Domestic FI SS	112	0.0275	0.1366	-0.0159	-0.2358	2.4104	-0.1499	0.1864	0.0299
	1277	(0.33)	(0.55)	(-0.40)	(-1.75)	(1.56)	(-0.56)	(0.70)	(0.18)

Table 14: Private equity and real estate – pension fund characteristics and performance

In Panel A in the first step we regress the private equity security selection (SS) return component (net benchmark-adjusted returns) on a four factor model that includes the Nasdaq excess returns, SMB, HML and LIQ. In Panel C in the first step we regress the real estate security selection (SS) return component (net benchmark-adjusted returns) on a four factor model that includes the MKT, SMB, HML and LIQ. In Panels B and D we add the momentum factor to the risk-adjusting models. We run these regressions for every fund that has at least 7 observations. In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). The following characteristics are included in the Fama-MacBeth regressions: LogSize – logarithm of total fund holdings, LogMand – logarithm of the total private equity holdings in Panels A and B (real estate holdings in panels C and D) and Costs – costs for investing in private equity in Panels A and B (real estate holdings in panels C and D). Mand_Liq is an interaction term of the log mandate size with the first step fund-specific loading on the liquidity factor. The first column # Funds and # Obs. present the number of funds (cross-sectional units) and the number of observations included in the analysis. In parentheses we report the t-statistics for every coefficient.

		Model 1	Model 2		Model 3		Model 4			Model 5		Model 6			
# Funds	s / # Obs.	Cons	Cons	LogSize	Cons	LogMand	Cons	LogMand	Mand_Liq	Cons	Costs	Cons	LogMand	Costs	Mand_Liq
Panel A: Pri	ivate Equi	ty without ad	ljusting for 1	nomentum i	in the first s	tep									
All Funds	122	5.8101	5.0884	0.0611	3.5332	0.4725	2.3750	0.2183	-1.7564	9.0040	-1.4698	8.0795	-0.2012	-1.6384	-1.8337
	1396	(3.95)	(2.54)	(0.17)	(2.63)	(1.62)	(1.83)	(0.82)	(-3.99)	(6.79)	(-3.64)	(5.48)	(-0.76)	(-4.07)	(-4.37)
U.S.	93	6.1515	7.0347	-0.1109	6.5277	0.0280	4.1774	-0.2585	-1.9778	9.3953	-1.4858	11.1325	-0.8485	-1.6499	-2.0799
	1067	(3.08)	(5.90)	(-0.35)	(2.20)	(0.06)	(1.83)	(-0.60)	(-4.58)	(4.72)	(-4.50)	(3.52)	(-1.72)	(-3.90)	(-5.33)
Canada	29	4.2622	3.4799	0.0762	0.9543	0.7390	-0.4966	1.3415	-1.7133	6.8633	-0.7525	6.2493	0.5258	-1.7362	-1.8370
	329	(3.95)	(1.10)	(0.24)	(0.32)	(1.47)	(-0.15)	(2.15)	(-2.75)	(8.52)	(-1.11)	(1.38)	(0.52)	(-1.34)	(-3.26)
					1 01										
Panel B: Pri	ivate Equi	ty with adjus	sting for mor	nentum in ti	he first step	1 2 2 2 2	< 1505	1 10 75	0.000	0 === 1	1	4 44 65	0.0045	1 20 55	1.0.100
All Funds	122	0.2651	-2.9278	0.3794	-5.2398	1.2232	-6.1797	1.1267	-0.9992	3.7754	-1.5634	-1.4167	0.8347	-1.3957	-1.0438
TT G	1396	(0.23)	(-0.74)	(0.87)	(-2.06)	(2.70)	(-2.88)	(2.55)	(-4.42)	(2.05)	(-2.05)	(-0.36)	(1.49)	(-1.60)	(-5.99)
U.S.	93	-0.0922	-2.2576	0.2420	-5.3424	1.1284	-6.9459	1.1898	-0.4735	4.2204	-1.9956	0.2279	0.7200	-2.0881	-0.6507
a 1	1067	(-0.06)	(-0.51)	(0.57)	(-1.26)	(1.72)	(-1.57)	(1.56)	(-1.12)	(2.89)	(-2.65)	(0.05)	(0.87)	(-2.01)	(-1.83)
Canada	29	0.7791	-16.6292	2.2014	-8.7594	2.6341	-11.2415	3.7131	-3.9379	1.7892	1.0724	-12.8786	3.9975	2.1789	-3.7712
	329	(0.58)	(-1.67)	(2.02)	(-1.96)	(2.84)	(-2.70)	(3.50)	(-3.45)	(0.42)	(0.56)	(-1.66)	(2.77)	(1.10)	(-3.44)
Panel C: Re	al Estate v	vithout adiu	sting for mo	mentum in t	he first sten										
All Funds	186	-0 4792	-3 3492	0 3466	-2.5550	0 4186	-2.9497	0.6177	-1 6864	-0.0520	-0 5438	-2 1954	0 5730	-0 6981	-1 7049
i in i unus	2164	(-1.67)	(-4.71)	(3.40)	(-6.92)	(5 35)	(-10.61)	(9.87)	(-9.07)	(-0.07)	(-0.90)	(-3.44)	(9.06)	(-1, 22)	(-9.98)
U.S.	123	-0.5368	-1.6259	0.1092	-2.3244	0.3172	-0.5302	0.1989	-2.0259	-0.6678	0.2531	-0.5751	0.2382	-0.3218	-1.9584
	1417	(-2.13)	(-0.73)	(0.39)	(-3.36)	(2.32)	(-0.25)	(0.65)	(-4.45)	(-0.71)	(0.27)	(-0.50)	(1.10)	(-0.46)	(-4.96)
Canada	63	-0.3418	-5.3507	0.6814	-3.7462	0.8111	-4.1493	0.8752	0.5426	1.0701	-2.9169	-2.3892	0.8733	-2.1019	-1.8626
	747	(-0.74)	(-2.69)	(3.04)	(-4.42)	(5.15)	(-4.84)	(3.96)	(0.20)	(1.16)	(-3.22)	(-1.96)	(5.42)	(-2.14)	(-3.35)
															(/
Panel D: Re	eal Estate v	vith adjustin	g for momen	ntum in the	first step										
All Funds	186	-0.2247	-4.1452	0.4744	-2.0854	0.3842	-2.3570	0.5036	-1.0573	0.4718	-0.9234	-1.2741	0.4343	-0.9964	-1.0772
	2164	(-0.88)	(-4.75)	(3.63)	(-5.11)	(4.11)	(-5.30)	(5.13)	(-5.28)	(0.73)	(-1.42)	(-1.46)	(3.95)	(-1.43)	(-5.40)
U.S.	123	-0.2407	-3.3706	0.3458	-1.1593	0.1680	-0.1555	0.1267	-1.4852	0.0174	-0.2413	-0.7276	0.2293	-0.3387	-1.3347
	1417	(-1.10)	(-1.35)	(1.18)	(-3.32)	(1.98)	(-0.10)	(0.57)	(-4.81)	(0.02)	(-0.23)	(-0.67)	(1.90)	(-0.26)	(-6.43)
Canada	63	-0.1493	-4.6605	0.6202	-3.5700	0.8294	-3.9280	1.1430	-1.8786	1.4210	-3.1841	-1.9369	0.7664	-2.3011	-0.6796
	747	(-0.33)	(-2.56)	(2.84)	(-4.73)	(6.03)	(-4.80)	(4.76)	(-1.62)	(2.20)	(-4.03)	(-2.15)	(5.17)	(-2.62)	(-1.63)

Table 15: Persistence in pension fund performance

This table presents the marginal effects after an ordered logit model. The dependent variable is the quintile ranking based on returns in year t+1 with 1 being lowest quintile ranking and 5 being the quintile with highest returns. The LY ranking independent variable is the quintile ranking in the previous year t. We also include the following variables: LogSize – log of average pension fund holdings in a given year, Costs – total fund costs, % Act – percentage of all holdings invested in active mandates and %Ext – percentage invested in external mandates. The marginal effects are estimated at the median values. In the ordered logit model we also add year dummy variables and cluster the standard errors by funds. Panel A presents the marginal effects for U.S. funds market timing returns, Panel B – U.S. funds security selection returns, Panel C – Canadian funds market timing returns and Panel D – Canadian funds security selection returns. The marginal effects are presented with their corresponding z-statistic in the parentheses.

Deulsine	Model 1	Model 2		Model 3		Model 4		
Ranking	LY ranking	LY ranking	Logsize	LY ranking	Costs	LY ranking	%Act	%Ext
Panel A: U	U.S. Funds Marke	et Timing Return	ls					
1	-0.0383	-0.0376	0.0073	-0.0373	-0.0909	-0.0364	-0.0048	-0.0538
	(-3.87)	(-3.86)	(1.22)	(-3.80)	(-2.31)	(-3.65)	(-0.14)	(-1.87)
2	-0.0099	-0.0100	0.0020	-0.0102	-0.0248	-0.0112	-0.0015	-0.0166
	(-1.37)	(-1.43)	(0.96)	(-1.47)	(-1.22)	(-1.69)	(-0.14)	(-1.13)
3	0.0077	0.0074	-0.0014	0.0073	0.0177	0.0063	0.0008	0.0094
	(1.15)	(1.11)	(-0.83)	(1.08)	(1.05)	(0.90)	(0.14)	(0.91)
4	0.0185	0.0183	-0.0036	0.0183	0.0445	0.0182	0.0024	0.0269
	(5.92)	(5.87)	(-1.26)	(5.78)	(2.45)	(5.59)	(0.14)	(1.88)
5	0.0219	0.0219	-0.0043	0.0220	0.0535	0.0230	0.0030	0.0340
	(3.15)	(3.16)	(-1.20)	(3.18)	(1.93)	(3.22)	(0.14)	(1.49)
Panel B: U	U.S. Funds Secur	ity Selection Ret	urns					
1	-0.0218	-0.0204	-0.0236	-0.0193	0.0324	-0.0196	0.0456	0.0790
	(-2.96)	(-2.86)	(-3.03)	(-3.10)	(0.79)	(-3.05)	(1.33)	(2.46)
2	-0.0071	-0.0061	-0.0070	-0.0090	0.0152	-0.0078	0.0181	0.0314
	(-1.48)	(-1.30)	(-1.39)	(-3.19)	(0.81)	(-2.70)	(1.35)	(2.33)
3	0.0034	0.0038	0.0044	0.0009	-0.0015	0.0022	-0.0051	-0.0088
	(0.72)	(0.88)	(0.86)	(0.29)	(-0.27)	(0.74)	(-0.64)	(-0.72)
4	0.0110	0.0102	0.0118	0.0105	-0.0176	0.0104	-0.0242	-0.0419
	(3.71)	(3.47)	(4.16)	(3.33)	(-0.80)	(3.38)	(-1.37)	(-2.60)
5	0.0144	0.0125	0.0145	0.0169	-0.0285	0.0148	-0.0345	-0.0597
	(2.30)	(2.19)	(2.54)	(3.34)	(-0.82)	(3.09)	(-1.39)	(-2.61)
Panel C: 0	Canadian Funds	Market Timing F	Returns					
1	-0.0243	-0.0200	0.0051	-0.0231	-0.1316	-0.0191	-0.0552	-0.0209
	(-3.70)	(-3.46)	(1.07)	(-3.49)	(-1.55)	(-3.33)	(-1.75)	(-1.07)
2	-0.0099	-0.0116	0.0030	-0.0097	-0.0551	-0.0118	-0.0340	-0.0129
	(-2.79)	(-3.34)	(1.07)	(-2.74)	(-1.42)	(-3.49)	(-1.70)	(-0.99)
3	0.0017	-0.0023	0.0006	0.0015	0.0084	-0.0031	-0.0088	-0.0033
	(0.56)	(-0.56)	(0.52)	(0.49)	(0.47)	(-0.75)	(-0.70)	(-0.56)
4	0.0124	0.0102	-0.0026	0.0119	0.0677	0.0096	0.0278	0.0105
	(3.88)	(3.01)	(-1.02)	(3.67)	(1.53)	(2.68)	(1.60)	(1.05)
5	0.0200	0.0237	-0.0061	0.0194	0.1105	0.0243	0.0703	0.0266
	(3.23)	(3.03)	(-1.08)	(3.07)	(1.50)	(3.09)	(1.67)	(0.98)
Panel D: 0	Canadian Funds	Security Selectio	on Returns					
1	-0.0398	-0.0374	-0.0132	-0.0429	0.0341	-0.0399	-0.0187	0.0896
	(-5.29)	(-5.00)	(-2.36)	(-4.12)	(0.32)	(-5.35)	(-0.42)	(2.72)
2	-0.0181	-0.0182	-0.0064	-0.0160	0.0127	-0.0157	-0.0073	0.0352
	(-4.29)	(-4.61)	(-2.15)	(-2.18)	(0.33)	(-3.58)	(-0.42)	(2.53)
3	0.0013	0.0002	0.0001	0.0046	-0.0037	0.0038	0.0018	-0.0085
	(0.25)	(0.05)	(0.05)	(0.50)	(-0.27)	(0.77)	(0.37)	(-0.75)
4	0.0210	0.0199	0.0070	0.0220	-0.0174	0.0208	0.0097	-0.0467
	(4.97)	(4.56)	(2.34)	(5.21)	(-0.32)	(5.37)	(0.42)	(-2.84)
5	0.0356	0.0355	0.0125	0.0323	-0.0256	0.0310	0.0145	-0.0696
	(5.09)	(5.18)	(2.10)	(3.14)	(-0.33)	(4.90)	(0.42)	(-2.70)



Figure 1: Asset allocation of U.S. and Canadian funds





Figure 2: Asset allocation of U.S. funds within equity, fixed income and alternatives

20% 10% 0%

TAA



Figure 3: Asset allocation of Canadian funds within equity, fixed income and alternatives



Appendix Table A.1: Time-series and cross-sectional return variation (related to Table 5)

This table reports the time-series and cross-sectional summary statistics on a fund level, incorporating all assets. Panel A displays the summary statistics from the cross-sectional distribution of R-squared statistics obtained from performing the following regression for each pension fund over time:

$$R_{i,t} - PR_t = \alpha_i + b_i R_{i,t}^k + \varepsilon_{i,t} , \quad i = 1, \dots, n$$

where $R_{i,t}$ is the net return of pension fund *i* at time *t* and PR_t is the average of equally weighted policy return for year *t*. $R_{i,t}^k$ refers to asset allocation return component $R_{i,t}^{AA}$, the market timing component $R_{i,t}^{MT}$ and the security selection component $R_{i,t}^{SS}$. The three different return components are defined in section 3. Panel B reports the summary statistics from the time-series distribution of R-squared statistics from the 19 (1990–2008) cross-sectional regressions:

$$R_{i,t} - PR_t = \alpha_i + b_i R_{i,t}^k + \varepsilon_{i,t}, \quad t = 1, \dots, T$$

where k = AA, MT, SS. At least four, seven or nine data points per fund are required to run each time-series or cross-sectional regression. In table 5 we required at least five data points per fund.

Table 5 with at least 4 data points per fund:

		All Funds			U.S.			Canada	
	Mean	Median	SD	Mean	Median	SD	Mean	Median	SD
Panel A: Time-series R-se	quared val	ues							
Asset Allocation policy	60.04	67.02	28.94	42.23	39.49	30.19	35.29	31.71	28.85
Market Timing	21.25	10.37	24.77	21.57	11.42	24.30	20.14	10.37	24.84
Security Selection	34.18	28.88	26.61	55.34	57.44	28.89	59.39	66.64	29.11
Panel B: Cross-sectional	R-squarea	l values							
Asset Allocation policy	50.92	52.77	21.87	35.18	35.80	16.75	23.99	20.78	13.49
Market Timing	4.53	2.49	4.97	4.16	3.77	3.48	7.62	4.23	10.67
Security Selection	33.11	37.29	21.66	45.59	46.12	18.70	48.81	46.65	19.78

Table 5 with at least 7 data points per fund:

		All Funds			U.S.			Canada	
	Mean	Median	SD	Mean	Median	SD	Mean	Median	SD
Panel A: Time-series R-se	quared val	ues							
Asset Allocation policy	59.09	63.46	27.54	40.03	37.59	29.14	32.22	27.75	27.34
Market Timing	14.41	7.13	17.74	15.72	8.71	19.36	12.45	5.18	15.37
Security Selection	33.69	30.04	24.98	54.54	55.78	27.89	57.82	66.50	28.02
Panel B: Cross-sectional	R-squarea	l values							
Asset Allocation policy	52.13	57.15	22.72	36.11	32.88	18.98	24.74	28.12	13.57
Market Timing	4.31	2.09	5.07	4.10	3.53	4.09	7.54	4.28	10.23
Security Selection	32.23	31.95	21.84	44.44	45.67	19.76	47.70	46.77	18.45

Table 5 with at least 9 data points per fund:

		All Funds			U.S.			Canada	
	Mean	Median	SD	Mean	Median	SD	Mean	Median	SD
Panel A: Time-series R-se	quared val	ues							
Asset Allocation policy	59.27	61.14	26.25	40.43	37.85	29.39	30.56	27.93	26.65
Market Timing	11.69	5.83	15.13	12.12	7.41	14.55	10.69	5.11	13.30
Security Selection	32.58	28.60	23.66	51.33	53.70	27.01	59.17	66.64	26.64
Panel B: Cross-sectional	R-squarea	l values							
Asset Allocation policy	54.47	61.89	22.81	39.93	37.80	19.72	25.38	25.91	13.23
Market Timing	4.16	1.67	5.05	3.84	3.06	3.86	7.69	3.48	11.72
Security Selection	31.56	30.93	22.42	42.47	41.71	20.33	48.72	48.31	18.80

Appendix Table A.2: Risk-adjusted performance per return components in all asset classes on a fund level (related to Table 7)

Robustness check - every fund included in the regressions below has at least 13 observations.

This table reports the net risk-adjusted performance at the fund level for all assets using a random coefficients. The following factors are included in the regressions: MKT, SMB and HML are the Fama-French factor returns, MOM – momentum factor, LIQ – Pastor and Stambaugh (2003) traded liquidity factor, FIMKT – fixed income excess market return. Dummy Y2000 is a dummy variable for year 2000 (Nortel effect). In Panel A the dependent variable is the return due to changes in asset allocation policy, which is calculated as the return due to changes in the strategic asset allocation weights in year *t* compared to year *t*-1 multiplied with the benchmark return $R_{i,t}^{CAA} = \sum_{j=1}^{N} (w_{i,j,t}^{AA} - w_{i,j,t-1}^{A})r_{i,j,t}^{BM}$. In Panel B the dependent variable is the market timing component of fund returns $R_{i,t}^{MT} = \sum_{j=1}^{N} (w_{i,j,t} - w_{i,j,t}^{AA})r_{i,j,t}^{BM}$, where $w_{i,j,t}^{BM}$ is the policy weight for fund *i* for asset class *j* and year *t*, $w_{i,j,t}$ is the actual realized weight for fund *i* for asset class *j* and year *t* and $r_{i,j,t}^{BM}$ is the benchmark return for fund *i* for the asset class *j* and period *t*. In Panel C we use the security selection component of fund returns as dependent variable $R_{i,t}^{SS} = \sum_{j=1}^{N} w_{i,j,t} (r_{i,j,t} - r_{i,j,t}^{BM})$, where $r_{i,j,t}$ is the realized net return on the asset class *j* for the year *t* by fund *i*. We report the annual alpha (Cons.) and betas with corresponding z-statistics in parentheses. RMSE is the root mean square error.

	# Funds	Cons.	MKT	SMB	HML	MOM	FIMKT	LIQ	Dummy	RMSE
	# Obs.								Y2000	
Panel A: Ch	anges in As	set Allocati	on return c	omponent: (Year t weig	hts – Year	t-1 weights) * Benchm	ark returns	
All Funds	107	0.1901	-0.0089	0.0025	-0.0085		0.0063	-0.0045		12.0859
	1630	(3.13)	(-3.44)	(1.04)	(-3.75)		(0.88)	(-1.19)		
All Funds	107	0.2047	-0.0084	0.0021	-0.0089	0.0003	0.0057	-0.0061		12.0819
	1630	(2.70)	(-2.83)	(0.82)	(-3.31)	(0.10)	(0.75)	(-1.46)		
U.S.	60	0.2814	-0.0102	0.0056	-0.0108		0.0063	-0.0112		13.4111
	896	(3.61)	(-2.88)	(1.45)	(-3.64)		(0.59)	(-2.51)		
U.S.	60	0.3306	-0.0106	0.0046	-0.0129	-0.0030	0.0088	-0.0116		13.4086
	896	(2.96)	(-2.47)	(1.11)	(-3.21)	(-0.65)	(0.78)	(-2.53)		
Canada	47	0.0709	-0.0082	-0.0001	-0.0081		0.0102	0.0044		10.3304
	734	(0.76)	(-2.13)	(-0.03)	(-2.28)		(1.07)	(0.73)		
Canada	47	0.0719	-0.0069	-0.0006	-0.0068	0.0022	0.0066	0.0013		10.3348
	734	(0.73)	(-1.70)	(-0.23)	(-1.93)	(0.72)	(0.66)	(0.18)		
Panel B: Ma	rket Timing	g return con	nponent: (A	ctual weigh	ts – Policy	weights) * I	Benchmark	returns		
All Funds	139	0.2515	-0.0028	0.0031	-0.0044		0.0084	0.0006		11.9486
	2196	(4.88)	(-1.18)	(1.28)	(-2.10)		(1.38)	(0.18)		
All Funds	139	0.2504	-0.0027	0.0025	-0.0041	0.0001	0.0089	0.0002		11.9543
	2196	(3.83)	(-1.05)	(0.95)	(-1.75)	(0.02)	(1.34)	(0.01)		
U.S.	81	0.2097	-0.0020	0.0060	-0.0052		0.0219	0.0020		13.1378
	1253	(3.19)	(-0.61)	(1.53)	(-1.79)		(2.78)	(0.50)		
U.S.	81	0.2291	-0.0027	0.0049	-0.0053	-0.0020	0.0246	0.0022		13.1516
	1253	(2.34)	(-0.78)	(1.12)	(-1.51)	(-0.47)	(2.73)	(0.54)		
Canada	58	0.3182	-0.0038	0.0003	-0.0029		-0.0076	-0.0011		10.2145
	943	(3.76)	(-1.11)	(0.17)	(-0.96)		(-0.81)	(-0.23)		
Canada	58	0.3044	-0.0025	-0.0000	-0.0022	0.0025	-0.0103	-0.0047		10.2123
	943	(3.66)	(-0.64)	(-0.01)	(-0.75)	(0.87)	(-1.11)	(-0.84)		
Panel C: Sec	curity Select	tion return	component:	Actual wei	ghts * (Rea	lized net re	turn – Benc	hmark retu	rn)	
All Funds	139	-0.0313	0.0112	0.0182	0.0112		-0.0296	0.0055	3.0524	11.8948
	2166	(-0.27)	(2.43)	(3.03)	(1.76)		(-2.08)	(0.82)	(8.15)	
All Funds	139	-0.5922	0.0234	0.0327	0.0334	0.0463	-0.0749	-0.0037	2.7448	11.8615
	2166	(-4.63)	(4.94)	(5.38)	(5.19)	(7.14)	(-4.58)	(-0.57)	(6.94)	
U.S.	81	0.2238	0.0135	0.0278	0.0212		-0.0637	-0.0094		12.9975
	1245	(1.38)	(2.13)	(2.87)	(2.52)		(-2.82)	(-1.18)		
U.S.	81	-0.8830	0.0368	0.0625	0.0591	0.0790	-0.1352	-0.0078		12.9551
	1245	(-4.66)	(5.46)	(6.62)	(6.60)	(8.29)	(-5.64)	(-1.04)		
Canada	58	-0.0347	0.0014	0.0006	0.0297		-0.0060	0.0077	3.2666	10.2986
	921	(-0.19)	(0.21)	(0.13)	(3.20)		(-0.30)	(0.66)	(6.36)	
Canada	58	-0.3472	0.0077	0.0031	0.0308	0.0235	-0.0212	-0.0068	3.9086	10.2801
	921	(-1.84)	(1.17)	(0.65)	(3.38)	(3.43)	(-1.06)	(-0.55)	(7.10)	

Appendix Table A.3: Risk adjusted returns per asset class (related to Table 8)

This table reports the net risk-adjusted performance in equity using a random coefficients model. Market timing (MT-E) and Security selection returns (SS-E) within Equity are risk-adjusted using the following factors: MKT, SMB and HML are the Fama-French factor returns, MOM – momentum factor, LIQ – Pastor and Stambaugh (2003) traded liquidity factor. Compared to Table 8 we do not control for year dummy 2000 ("Nortel effect"). Domestic equity SS-DE column displays the security selection (net benchmark-adjusted returns) for U.S. Funds mandates in U.S. equity and Canadian Funds mandates in Canadian equity. The table shows the alpha, its corresponding z-statistic and the root mean square error (RMSE) from all regressions.

	Equ	uity	Domestic Equity
	MT-E	SS-E	SS-DE
Panel A: All	Funds without	Year dummy 20	00
# Funds	287	287	285
# Obs.	3504	3470	3440
Alpha	0.2079	-0.1793	0.5261
	(4.12)	(-0.89)	(2.11)
RMSE	12.0470	12.0894	12.5198
Panel B: Ca	nada without Ye	ar dummy 2000)
# Funds	111	111	110
# Obs.	1409	1392	1380
Alpha	0.1808	0.8014	2.5456
	(2.54)	(2.83)	(6.60)
RMSE	10.1103	10.2763	11.1858

MT: (Actual weights – Policy weights) * Benchmark returns SS: Actual weights * (Realized net return – Benchmark return)

Appendix Table A.4: Pension fund characteristics and total market timing (MT) returns (related to Table 10)

In the first step we regress the total market timing returns on a five factor that includes the MKT, SMB, HML, LIQ and FIMKT. In Panels B and D we also add the momentum factor to the risk-adjusting model. We run these regressions for every fund that has at least 8 observations, which results in 256 Funds (3297 observations) in All funds models, 152 U.S. Funds (1937 observations) and 104 Canadian funds (1360 observations). In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run a Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). We include the following characteristics: SizeAct – log of total holdings in all passive mandates, SizeExt – log of total holdings in all external mandates, SizeInt – log of total holdings in all internal mandates, LogSize – log of average pension fund holdings in a given year, %Act – percentage of all holdings invested in active mandates, %Ext – percentage of all holdings invested in external mandates and Costs – total fund costs. SizeLiq is an interaction term of the log fund size with the first step fund-specific loading on the liquidity factor. In parentheses we report the t-statistics for every coefficient.

	Model 1			Model 2			-				
	Cons.	SizeAct	SizePas	Cons.	SizeExt	SizeInt	-				
Panel A: wi	ithout mom	entum factor	r in the first	step			_				
All Funds	0.2505	0.0059	-0.0048	0.2833	0.0003	-0.0006					
	(4.96)	(0.75)	(-1.81)	(4.37)	(0.03)	(-0.13)					
U.S.	0.5326	-0.0257	-0.0055	0.4253	-0.0120	-0.0112					
	(5.37)	(-1.94)	(-1.36)	(7.25)	(-1.27)	(-1.97)					
Canada	0.1526	0.0209	-0.0104	0.4520	-0.0399	0.0157					
	(1.43)	(1.88)	(-1.31)	(1.56)	(-0.91)	(2.30)					
Panel B: wi	ith momenti	um factor in	the first ste	р			_				
All Funds	0.3184	-0.0037	0.0046	0.4236	-0.0140	0.0006					
	(3.11)	(-0.37)	(0.90)	(7.16)	(-1.47)	(0.09)					
U.S.	0.3868	-0.0090	0.0052	0.5290	-0.0217	0.0003					
	(2.25)	(-0.71)	(0.92)	(5.97)	(-2.53)	(0.03)					
Canada	0.4918	-0.0311	-0.0109	0.8732	-0.0982	0.0065					
	(3.41)	(-1.88)	(-1.34)	(2.54)	(-2.05)	(0.89)	-				
	Model 3					Model 4					
	Cons	LogSize	Costs	%Act	%Ext	Cons	LogSize	SizeLiq	Costs	%Act	%Ext
Panel C: w	ithout adjus	sting for mor	nentum in t	he first step							
All Funds	0.3069	-0.0061	0.4067	-0.0091	-0.1043	0.4054	-0.0202	-1.2572	0.1855	0.0846	-0.0898
	(2.74)	(-1.38)	(3.37)	(-0.07)	(-1.85)	(2.14)	(-2.03)	(-12.22)	(1.19)	(0.94)	(-1.59)
U.S.	0.2126	-0.0145	0.5884	0.0162	0.0170	0.4048	-0.0258	-1.3631	0.3012	0.0731	-0.0568
	(1.14)	(-0.89)	(3.46)	(0.12)	(0.19)	(4.11)	(-2.30)	(-10.07)	(1.75)	(0.97)	(-1.11)
Canada	1.0166	-0.0782	-0.7932	0.1631	-0.2737	0.8018	-0.0705	-1.5137	-0.7988	0.2796	-0.1516
	(1.19)	(-0.98)	(-1.23)	(0.84)	(-1.73)	(0.87)	(-0.82)	(-15.60)	(-1.17)	(1.77)	(-0.83)
Panel D: w	ith adjustin	g for momer	ntum in the j	first step							
All Funds	0.5620	-0.0054	0.5641	-0.2030	-0.2334	0.1931	0.0146	-1.3897	0.5969	-0.0278	-0.1614
	(6.05)	(-0.96)	(4.19)	(-1.51)	(-2.77)	(1.28)	(1.77)	(-7.46)	(3.05)	(-0.28)	(-3.13)
U.S.	0.6117	0.0028	1.0418	-0.3590	-0.3909	0.5115	0.0057	-2.0635	0.6782	0.0186	-0.4203
	(3.36)	(0.19)	(5.47)	(-2.21)	(-6.47)	(2.81)	(0.34)	(-7.97)	(2.08)	(0.28)	(-8.05)
Canada	1.5300	-0.1406	-0.7330	0.1719	-0.3910	1.2224	-0.1561	-1.2775	-0.4630	0.4126	-0.2638
	(2.40)	(-1.96)	(-1.17)	(1.33)	(-2.41)	(1.80)	(-2.15)	(-15.24)	(-0.65)	(3.93)	(-2.21)

Appendix Table A.5: Pension fund characteristics and total security selection (SS) returns (related to Table 11)

In the first step we regress the total security selection returns on a five factor that includes the MKT, SMB, HML, LIQ and FIMKT. In Panels B and D we also add the momentum factor to the risk-adjusting model. The regressions for all funds and Canada contain also year dummy 2000. We run these regressions for every fund that has at least 8 observations (9 observations if we include year dummy 2000 in the first step), which results in 224 Funds (3044 observations) in All funds models, 152 U.S. Funds (1937 observations) and 88 Canadian funds (1235 observations). In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run a Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). We include the following characteristics: SizeAct – log of total holdings in all active mandates, SizePas – log of total holdings in all external mandates, SizeInt – log of total holdings in all internal mandates, LogSize – log of average pension fund holdings in a given year, %Act – percentage of all holdings invested in active mandates, %Ext – percentage of all holdings invested in external mandates and Costs – total fund costs. SizeLiq is an interaction term of the log fund size with the first step fund-specific loading on the liquidity factor. In parentheses we report the t-statistics for every coefficient.

	Model 1			Model 2			_				
	Cons.	SizeAct	SizePas	Cons.	SizeExt	SizeInt	-				
Panel A: wi	ithout mom	entum factor	\cdot in the first	step			-				
All Funds	-0.1933	0.0248	0.0156	-0.0187	0.0032	0.0334	-				
	(-1.24)	(1.22)	(1.32)	(-0.17)	(0.17)	(8.12)					
U.S.	0.5771	-0.0186	0.0125	0.6239	-0.0148	0.0118					
	(2.01)	(-0.53)	(0.79)	(3.49)	(-0.45)	(0.88)					
Canada	-0.3653	0.0436	0.0113	0.5385	-0.1049	0.0623					
	(-2.10)	(3.62)	(0.68)	(1.93)	(-2.14)	(9.98)	_				
Panel B: wi	ith momenti	um factor in	the first ste	p			_				
All Funds	-0.2919	-0.0303	-0.0226	0.1067	-0.1062	0.0493					
	(-2.09)	(-1.59)	(-1.85)	(0.34)	(-3.47)	(5.51)					
U.S.	-2.3509	0.1568	0.0168	-1.3441	0.0357	0.0602					
	(-8.36)	(3.58)	(0.96)	(-4.70)	(1.35)	(6.76)					
Canada	-0.4154	0.0161	0.0011	0.5481	-0.1337	0.0257					
	(-1.25)	(0.36)	(0.08)	(3.79)	(-10.12)	(1.58)	-				
	Model 3					Model 4					
	Cons	LogSize	Costs	%Act	%Ext	Cons	LogSize	SizeLiq	Costs	%Act	%Ext
Panel C: w	ithout adjus	sting for mor	nentum in ti	he first step				•			
All Funds	0.2028	0.0409	0.9546	-0.3952	-0.4862	0.4096	0.0316	-1.3285	0.3037	-0.1510	-0.5589
	(0.54)	(1.30)	(3.60)	(-1.52)	(-2.76)	(1.01)	(1.06)	(-11.52)	(1.67)	(-0.51)	(-3.55)
U.S.	-0.4544	0.0630	0.7544	-0.0348	0.3237	-0.2556	0.0386	-1.4938	0.0712	0.0349	0.2844
	(-0.62)	(1.09)	(0.66)	(-0.06)	(0.80)	(-0.36)	(0.68)	(-5.48)	(0.06)	(0.06)	(0.68)
Canada	1.2482	-0.0179	1.0275	-0.6387	-1.0275	1.7419	-0.0425	-1.2132	-0.2710	-0.4181	-1.0729
	(1.38)	(-0.21)	(1.06)	(-4.08)	(-4.10)	(1.99)	(-0.48)	(-18.26)	(-0.28)	(-1.54)	(-4.64)
Panel D: w	ith adjustin	g for momen	ntum in the j	first step							
All Funds	1.4511	-0.1308	-0.5096	-0.1606	-0.9257	1.6887	-0.1540	-0.8496	-0.9246	-0.0482	-0.9492
	(3.86)	(-5.62)	(-1.58)	(-0.73)	(-3.65)	(5.10)	(-6.06)	(-5.13)	(-2.02)	(-0.23)	(-3.91)
U.S.	-2.0231	0.1642	-1.1994	0.2196	-0.0800	-1.5784	0.1000	-0.6423	-1.6881	0.3766	0.0071
	(-2.58)	(3.91)	(-1.52)	(0.41)	(-0.18)	(-2.31)	(1.94)	(-2.08)	(-1.73)	(0.82)	(0.02)
Canada	0.4655	-0.0032	1.9715	-0.3471	-0.9923	0.8593	-0.0242	-0.6976	1.1510	-0.4823	-0.8883
	(0.40)	(-0.03)	(0.97)	(-1.86)	(-3.42)	(0.76)	(-0.24)	(-5.22)	(0.50)	(-2.74)	(-4.52)

Appendix Table A.6: Pension fund characteristics and security selection (SS) returns (related to Table 11)

In this table we do not control for "Nortel effect" and it can be compared with Table 11 and Appendix Table A.5

In the first step we regress the total security selection (net benchmark-adjusted) returns on a six factor model that includes the MKT, SMB, HML, MOM, LIQ and FIMKT. We run these regressions for every fund that has at least 8 observations, which results in 256 Funds (3297 observations) in Panel A, and 104 Funds (1360 observations) in Panel B. In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run a Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). We include the following characteristics: Log(Size) – log of average pension fund holdings in a given year, %Act – percentage of all holdings invested in active mandates, %Ext – percentage of all holdings invested in External Mandates and Costs – total fund costs. We also include four other size-related variables: SizeAct – log of total holdings in all active mandates, SizePas – log of total holdings in all external mandates and SizeInt – log of total holdings in all passive mandates, SizeExt – log of total holdings in all external mandates and SizeInt – log of total holdings in all passive mandates, SizeExt – log of total holdings in all external mandates and SizeInt – log of total holdings in all passive mandates, SizeExt – log of total holdings in all external mandates and SizeInt – log of total holdings in all passive mandates, SizeLiq is an interaction term of the log fund size with the first step fund-specific loading on the liquidity factor. Panel A presents the results for both U.S. and Canadian funds, while in Panel B we show the results of Canadian funds only. In parentheses we report the t-statistics for every coefficient.

Cons.	Log(Size)	SizeLiq	Costs	%Act	%Ext	SizeAct	SizePas	SizeExt	SizeInt
Panel A:	· All Funds								
-0.1213									
(-0.90)									
1.9874	-0.2648								
(3.22)	(-4.38)								
1.6149	-0.2400	-1.0931							
(3.17)	(-4.67)	(-4.08)							
2.1123								-0.2997	0.0507
(2.75)								(-3.70)	(3.23)
1.5523						-0.1869	-0.0590		
(2.45)						(-2.92)	(-4.62)		
0.6335			-2.5326						
(3.32)			(-13.67)						
3.5599	-0.3544	-1.1474	-3.7664						
(5.07)	(-5.67)	(-4.33)	(-5.37)						
0.0142				0.9779	-1.1034				
(0.03)				(2.79)	(-3.30)				
4.3207	-0.4133		-3.2000	1.0467	-1.2516				
(4.10)	(-5.33)		(-5.67)	(3.98)	(-2.67)				
3.8806	-0.3918	-1.1317	-3.5442	1.0038	-1.0277				
(5.24)	(-6.18)	(-4.42)	(-4.65)	(4.83)	(-3.25)				
Panel B:	· Canada								
1.0898									
(5.30)									
3.4071	-0.3161								
(3.88)	(-3.84)								
2.5959	-0.2452	-1.0734							
(3.79)	(-3.86)	(-4.60)							
5.0383								-0.5614	0.0137
(3.90)								(-3.68)	(0.44)
2.8109						-0.2042	-0.0813		
(4.03)						(-4.68)	(-2.50)		
1.1131			0.1260						
(3.26)			(0.14)						
2.7897	-0.2409	-1.1476	-0.7368						
(2.12)	(-1.97)	(-4.54)	(-0.43)						
0.5803				1.4269	-0.7636				
(1.03)				(2.62)	(-1.55)				
5.8969	-0.5463		0.4446	1.3343	-2.2656				
(2.59)	(-2.56)		(0.30)	(2.51)	(-2.75)				
4.3427	-0.4208	-1.0919	-0.4270	1.1763	-1.4614				
(3.11)	(-2.61)	(-4.51)	(-0.22)	(1.89)	(-3.20)				

Appendix Table A.7: Replication of Dyck and Pomorski (2011)

This table can be compared with Table 3 of Dyck and Pomorski (February 2011). The dependent variable is the overall fund net benchmark-adjusted return in year t (security selection return component on a fund level). The main independent variable is the log of year t-1 fund size. Regressions are estimated over the pooled sample of U.S. and Canadian funds (All) or on a single-country level and, where indicated, we use also year fixed effects. Corporate is a dummy variable, which is equal to 1 if the pension fund is classified as corporate and 0 otherwise.

	U.S.	U.S.	U.S.	Canada	All
Log of end of year t-1 plan size	0.1083	0.0895	0.1071	0.0683	0.0864
	(2.28)	(1.97)	(2.31)	(1.50)	(2.65)
Corporate plan dummy			0.2681	0.1791	0.2210
			(1.98)	(1.38)	(2.22)
Observations	2175	2175	2175	1393	3568
R-squared	0.0024	0.1436	0.1451	0.2360	0.1184
Year FE	NO	YES	YES	YES	YES
Plan FE	NO	NO	NO	NO	NO

Appendix Table A.8: Equity – pension fund characteristics and performance (related to Table 12)

In the first step we regress the equity security selection (SS) (net benchmark-adjusted returns) or market timing (MT) return component on a four factor model that includes the MKT, SMB, HML and LIQ. We run these regressions for every fund that has at least 7 observations. For Canadian funds here we do not add year dummy 2000 to the factor model. In Panel B we also add MOM – momentum factor to the model. In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). The following characteristics are included in the Fama-MacBeth regressions: LogMand – log of the total equity holdings, Costs – costs for investing in equity, %ActE – percentage in active mandates and %ExtE – percentage in external mandates from the equity holdings. For U.S. Small Cap %ActE and %ExtE are estimated based on assets in U.S. small cap equity. Mand_Liq is an interaction term of the log mandate size with the first step fund-specific loading on the liquidity factor. The first column # Funds and # Obs. present the number of funds and the number of observations included in the analysis. In parentheses we report the t-statistics for every coefficient.

	# Funds	Model 1	Model 2		Model 3			Model 4		Model 5		
	# Obs.	Cons.	Cons.	LogMand	Cons.	LogMand	Mand_Liq	Cons.	Costs	Cons	%ActE	%ExtE
Panel A: without momentu	т											
All Funds Equity SS no	287	0.7378	2.8938	-0.2860	1.4645	-0.1500	-1.4554	0.7354	0.0155	-0.2881	1.7303	-0.3595
year dummy	3470	(3.13)	(3.32)	(-3.25)	(2.47)	(-2.42)	(-7.80)	(3.04)	(0.02)	(-0.51)	(1.94)	(-2.03)
All Funds Equity SS	256	0.0810	0.3844	-0.0403	0.4016	-0.0459	-1.3731	0.4838	-1.4226	0.2935	0.2968	-0.5280
with year dum 2000	3253	(0.54)	(1.15)	(-0.93)	(1.03)	(-0.92)	(-6.29)	(2.28)	(-2.94)	(0.46)	(0.33)	(-4.56)
All Funds Equity MT	286	0.1677	0.1018	0.0100	0.2089	-0.0056	-0.8762	0.1702	-0.0157	0.2758	-0.0833	-0.0451
	3460	(4.47)	(0.77)	(0.55)	(1.76)	(-0.38)	(-8.06)	(5.72)	(-0.12)	(2.70)	(-0.53)	(-0.65)
All Funds Dom Equity	285	1.1968	5.1669	-0.5678	3.1238	-0.3375	-1.6681	1.0809	0.5095	-0.4036	2.3776	-0.1734
SS no year dummy	3440	(5.05)	(5.59)	(-5.56)	(5.62)	(-5.10)	(-10.56)	(7.36)	(1.02)	(-2.21)	(4.59)	(-0.80)
All Funds Dom Equity	249	0.3868	2.0369	-0.2363	1.7037	-0.1759	-1.5594	0.5906	-0.7431	0.1701	0.6684	-0.2976
SS with year dum 2000	3188	(2.80)	(7.17)	(-5.81)	(5.11)	(-3.83)	(-16.39)	(4.63)	(-0.68)	(0.84)	(10.08)	(-0.84)
Canada Equity SS no	111	1.0465	3.3224	-0.3371	2.2739	-0.2464	-1.0236	0.9354	0.4015	0.6857	1.1070	-0.5555
year dummy	1392	(3.23)	(2.94)	(-3.13)	(2.82)	(-3.42)	(-4.32)	(3.07)	(0.50)	(1.97)	(1.42)	(-2.07)
Canada Domestic Equity	110	2.2977	5.2507	-0.4948	3.7043	-0.3599	-1.5373	1.9023	2.0259	0.5231	2.1597	-0.0342
SS no year dummy	1380	(5.80)	(5.78)	(-6.08)	(6.20)	(-5.61)	(-4.99)	(4.13)	(2.29)	(1.30)	(2.90)	(-0.10)
Panel B: with momentum												
All Funds Equity SS no	287	-0.0670	2.6897	-0.3715	2.2219	-0.3393	-1.2818	0.5079	-2.0017	-0.3972	1.8779	-1.3026
year dummy	3470	(-0.39)	(3.58)	(-4.74)	(3.18)	(-4.52)	(-7.20)	(2.20)	(-3.34)	(-0.56)	(1.86)	(-3.92)
All Funds Equity SS	256	-0.7136	-0.2734	-0.0587	-0.1468	-0.0862	-1.2407	-0.0635	-2.3063	0.0054	0.4219	-1.2306
with year dum 2000	3253	(-6.45)	(-0.72)	(-1.27)	(-0.32)	(-1.45)	(-2.14)	(-0.44)	(-5.20)	(0.01)	(0.42)	(-5.79)
All Funds Equity MT	286	0.2605	0.2049	0.0089	0.3180	-0.0072	-0.5979	0.2067	0.1766	0.3787	-0.0505	-0.0814
	3460	(6.73)	(1.69)	(0.53)	(2.74)	(-0.49)	(-4.89)	(5.75)	(1.42)	(4.48)	(-0.35)	(-1.21)
All Funds Dom Equity	285	0.7160	5.3794	-0.6701	4.6192	-0.5892	-1.4591	1.3096	-2.4780	-0.6156	2.9185	-1.0083
SS no year dummy	3440	(3.73)	(6.86)	(-7.45)	(6.50)	(-6.81)	(-9.20)	(7.33)	(-6.48)	(-2.27)	(5.83)	(-3.35)
All Funds Dom Equity	249	0.0388	1.8743	-0.2618	1.7425	-0.2360	-1.0654	0.4781	-0.0171	-0.1149	1.1207	-0.8102
SS with year dum 2000	3188	(0.32)	(4.66)	(-5.11)	(3.75)	(-3.90)	(-3.39)	(2.51)	(-1.42)	(-0.41)	(4.50)	(-3.02)
Canada Equity SS no	111	1.0020	3.8010	-0.4181	3.5830	-0.3978	-0.9111	0.8032	0.8055	0.4479	1.2443	-0.4447
year dummy	1392	(3.06)	(3.40)	(-4.11)	(3.26)	(-3.81)	(-2.41)	(2.47)	(0.76)	(1.86)	(1.48)	(-1.95)
Canada Domestic Equity	110	2.6351	6.0636	-0.5734	5.0945	-0.4183	-1.7546	2.0425	3.1911	0.5872	2.2755	0.1875
SS no year dummy	1380	(6.90)	(6.32)	(-6.70)	(4.40)	(-2.76)	(-4.19)	(4.32)	(2.35)	(1.83)	(2.95)	(0.52)

Appendix Table A.9: Fixed income – pension fund characteristics and performance (related to Table 13)

In the first step we regress the fixed income security selection (SS) (net benchmark-adjusted returns) or market timing (MT) return component on a four factor model that includes the FIMKT, MKT, OPTION and HY. We run these regressions for every fund that has at least 6 observations. In the second step we augment the alphas retrieved from the first step with the error terms of the first step and run Fama-MacBeth regressions and correct for autocorrelation and heteroskedasticity (using Newey-West with three lags). The following characteristics are included in the Fama-MacBeth regressions: LogMand – logarithm of the total fixed income holdings, Costs – costs for investing in Fixed Income, %ActFI – percentage in active mandates from the fixed income holdings. The first column # Funds and # Obs. present the number of funds (cross-sectional units) and the number of observations included in the analysis. This table presents results of the joint analysis of U.S. and Canadian funds. In parentheses we report the t-statistics for every coefficient. Market timing component within fixed income requires that the fund invests in at least two types of fixed income (for example: Canadian fixed income and EAFE fixed income). In that case there can be a difference in weights within the fixed income, which will lead to return component that is due to the difference from actual and strategic weights.

		Model 1	Model 2		Model 3		Model 4		
	# Funds	Cons.	Cons.	LogMand	Cons.	Costs	Cons.	%ActFI	%ExtFI
	# Obs.			-					
FI SS	318	0.0221	0.4382	-0.0629	-0.1076	0.8411	-0.1909	0.1591	0.0916
	3661	(0.26)	(4.43)	(-6.77)	(-1.24)	(2.44)	(-2.03)	(1.01)	(2.08)
FI MT	232	0.0012	-0.0357	0.0051	-0.0171	0.1120	-0.0612	0.1000	-0.0250
	2814	(0.11)	(-1.01)	(0.83)	(-0.73)	(1.06)	(-2.18)	(2.56)	(-1.40)
Domestic FI SS	298	-0.0102	0.2162	-0.0344	-0.1603	1.0433	0.0532	0.0046	-0.0890
	3346	(-0.11)	(1.50)	(-1.64)	(-2.41)	(4.57)	(0.43)	(0.02)	(-0.92)

Appendix Table A.10: Comparison with Dyck and Pomorski (2011)

This table can be compared with Table 7 Panel B of Dyck and Pomorski (February 2011). The dependent variable is the net benchmark-adjusted return on private equity, real estate, hedge funds, equity, U.S. equity or fixed income. The coefficient presented refers to log of year t-1 holdings in the give asset class. Regressions are estimated over the pooled sample of U.S. and Canadian funds and, where indicated, we use also year or plan fixed effects. In model 2 we also add corporate dummy, which is equal to 1 if the pension fund is classified as corporate and 0 otherwise.

	(1)	(2)	(3)
Private Equity	1.5226	1.5703	1.0361
	(4.64)	(4.79)	(1.53)
Real Estate	0.5232	0.5471	0.1852
	(5.91)	(6.10)	(0.86)
Hedge Funds	0.1790	0.1713	-3.2489
	(0.53)	(0.51)	(-2.53)
Equity	-0.0214	-0.0088	0.2327
	(-0.54)	(-0.22)	(2.16)
U.S. Equity	-0.0184	-0.0121	-0.0537
	(-0.34)	(-0.22)	(-0.39)
Fixed Income	0.0609	0.0561	0.0091
	(2.98)	(2.67)	(0.17)
Corporate Dummy	NO	YES	NO
Year FE	YES	YES	YES
Plan FE	NO	NO	YES

Appendix Table A.11: Persistence in pension fund performance

In Panels A and C funds are placed into quintiles based on their market timing returns. In Panels B and D U.S. and Canadian funds are placed into quintiles based on their security selection (net benchmark-adjusted) returns. High row or column represents the quintile with the highest market timing return. Percentages represent the probability that a fund which was ranked in one of the 5 quintiles in year *t* ends up in one of the quintiles in year t+1 or exits the database. Exit column presents the percentage of funds exiting the CEM database in year t+1. Return in t+1 columns present the market timing or security selection returns in year t+1 of the top and bottom quintiles, which are formed in year *t*. Test Diff column is a t-statistic of the difference in returns between the low and high quintile.

Panel A: U.S. Funds Market Timing Returns													
			Return	in <i>t</i> +1	Test								
		Low	2	3	4	High	Exit	Low	High	Diff			
	Low	23.49%	16.78%	11.74%	13.26%	11.07%	23.66%	0.0506	0.3073	2.79			
Year t	2	14.77%	18.17%	17.15%	11.71%	10.87%	27.33%						
ranking	3	12.95%	16.35%	19.76%	15.33%	10.56%	25.04%						
	4	10.87%	15.62%	14.60%	19.86%	13.58%	25.47%						
	High	17.21%	9.64%	11.02%	14.11%	21.51%	26.51%						

Panel B: U.S. Funds Security Selection Returns

		Return in $t+1$		Test						
		Low	2	3	4	High	Exit	Low	High	Diff
	Low	20.54%	12.56%	13.41%	14.94%	12.56%	25.98%	-0.0912	0.6156	2.90
Year t	2	13.01%	15.92%	16.44%	15.75%	10.96%	27.91%			
ranking	3	11.74%	17.27%	16.93%	16.23%	13.47%	24.35%			
	4	12.50%	15.58%	17.64%	16.78%	14.55%	22.95%			
	High	14.63%	12.02%	12.37%	14.46%	20.38%	26.13%			

Panel C: Canadian Funds Market Timing Returns

		Return in $t+1$		Test						
		Low	2	3	4	High	Exit	Low	High	Diff
	Low	22.68%	17.49%	14.48%	13.39%	12.57%	19.40%	0.1434	0.2794	1.46
Year t	2	17.93%	17.65%	16.25%	14.29%	13.73%	20.17%			
ranking	3	11.90%	17.00%	22.66%	17.00%	12.75%	18.70%			
	4	14.37%	15.21%	16.34%	18.87%	17.18%	18.03%			
	High	18.29%	12.00%	11.43%	16.57%	19.43%	22.29%			

Panel D: Canadian Funds Security Selection Returns

		Return in $t+1$		Test						
		Low	2	3	4	High	Exit	Low	High	Diff
	Low	20.85%	19.44%	12.68%	11.27%	13.24%	22.54%	0.0227	0.9041	3.42
Year t	2	21.39%	13.87%	20.23%	14.45%	10.98%	19.08%			
ranking	3	16.09%	19.25%	17.82%	17.24%	10.63%	18.97%			
	4	11.85%	17.05%	15.90%	19.08%	17.34%	18.79%			
	High	14.20%	8.58%	12.43%	18.64%	24.85%	21.30%			